Installation and Operation Manual for the SP500 A-C V★S⁽³⁾ Controller Three-Phase Input and Three-Phase Output

1/4-5 HP @ 200-230 VAC, 1/4-10 HP @ 380-460 VAC and 1/4 -10 HP @ 575 VAC NEMA 1 NEMA 4X (Indoor Only)/12



Instruction Manual D2-3267-1



DANGER

ONLY QUALIFIED ELECTRICAL PERSONNEL FAMILIAR WITH THE CONSTRUCTION AND OPERATION OF THIS EQUIPMENT AND THE HAZ-ARDS INVOLVED SHOULD INSTALL, ADJUST, OPERATE, AND/OR SERVICE THIS EQUIPMENT. READ AND UNDERSTAND THIS MANUAL IN ITS ENTIRETY BEFORE PROCEEDING. FAILURE TO OBSERVE THIS PRECAUTION COULD RESULT IN SEVERE BODILY INJURY OR LOSS OF LIFE.

V★S≤ and Reliance≥ are registered trademarks of Reliance Electric Industria Company or its subsidiaries.

Copyright, Reliance Electric Industrial Company, 1993.

Table of Contents

1:	Receive and Accept Shipment	1-1
	About this Instruction Manual	1-1
	Identity the Controller	1-2
	Controller Description	1-3
	Controller Namepiete	1.5
	Receive and Accept the Shipment .	1-5
	File a Return Request	16
	Storage until Installation	1-8
	SP500 Denro Packages - Model 1SU14001 Only	1-6
2:	Specifications	2.1
	Controller Specifications	2-1
	Service Conditione	2-5
	Controller Input/putput Specifications	26
	Contro ler inputa	2-6
	Contro ler Outouts	27
	Controller Application Data	28
	Motor Applications	2-8
	Single Mater Applications	29
	Multi-Motor Applications	2-9
	Motor Lead Lengths	2-10
	Controller Optional Kits	2-11
	Controller Power Loss	2-12
3:	Installation and Wiring	3-1
	Planning and Location	3-1
	Controller Mounting	3-2
	Size A Controller Mounting (230VAC 182 HP)	3.2
	Size B Controller Mounting (460 VAC 1 5 HP & 575 VAC 1 5 HP)	34
	Size C Controller Mounting	
	(2S0 VAC 2 (NEMA 4), S and 5 HP 460 VAC 7.5 and 10 HP, and	
	575 VAC 7.5 and 10 HP(3-5
	Install an External Input Disconnect	3-6
	Install A-O Branch Circuit Protection	3-7
	Trans'ormer Installation (If Needed)	3-9
	Input Transformers	39
	Output Transformers	3-9
	Grounding	3-15
	Motor Preparation	3-19
	Controller Wiring	3-11
	Power Winng	3-15
	Control and Signal Wiring	3-18
	Ramola Analog hpu. Reference Jumpar Selling	3-25
	Uptional Jynamic Braking Winng	3 26

4:	Keypad and Display Operation	4-1
	Keyped and Display Description	4-1
	Indicator Lights	4-1
	Display Made LEDs	4.2
	Operation Mode LEDs	1-3
	Key Descriptions	4-4
	Power Up	4-7
	Modes of Conholler Operation	4.7
	Display Mtdrs	47
	Program Mode	4.8
	GrorLag	4 10
5:	Adjusting the Drive Functions	5-1
	FUNCTIONS	
	FOR DEMOTE ENABLE	5-4
	FAU - REMOTE ENABLE	3-3
		2-4
	F-02 - DEGELERATION	0-0
	F-03 - MINIMUM SPEED	5-6
	F 04 - MAXIMUM SPLLD	07
	F-05 - GURHFINT HMH	5-8
	F-06 - MANUAL TOHOOE BOOST	5-9
	F 07 - WHZ (Base Speed)	6 1D
	F-06 - RPM AT BASE SPEED	5-1-
	F-09 - TET (INSTANTANEOUS ELECTRONIC THIP)	
	QUIPUT ENABLE	5-12
	F-10 - CARHIER FREQUENCY	5-13
	F 11 - REMOTE REFERENCE GAIN	5 14
	F-12 - REMOTE REFERENCE OFFS-1	5-15
	F-13 - REMOTE REFERENCE DISPLAY ENABLE	
	(4TH DISPLAY MODE)	\$ 16
	F-14 ELECTRONIC THERMAL OVERLOAD	5-17
	F-15 - ELECTRONIC THERMAL OVERLOAD ENABLE	9-1R
	F-16 - GUAST STOP =NABLE	5-19
	F-17 - REVERSE DISABLE	5-20
	F-18 - VERSION INFORMATION	5-21
	ERH - ESHORIOG	5-72
6:	Start the Controller	8-1
	Check the Installation	6-1
	Sart the Controller	83
	Cover Installation, NEMA 4X (Indoor Only)/12	6-5
7:	Troubleshopting, Fault Codes, and Replacement Parts	7-1
	System Operation	7-1
	Fault Codes	1-4
	How to Access and Bead the Error Log	7-4
	Verily D-C Bue Voltage	7-5
8:	Glossary of Common Terms	8-1
9:	Index	9-1

List of Figures

Figure 1.1.	Controller Functional Block Diagram	14
Figure 1-2.	Namepiate Information.	1-5
Figure 2-1.	Matar Lead Length Connections	2 10
Figure 3-1.	Size A Physical Dimensions	3-3
Figure 3-2.	Size B Physical Dimensions	34
Figure 3-3.	Size C Physical Dimensions.	3-5
Figure 3-∠.	Size A Wiring Locations.	3-12
Figure 3-5.	Size H Wring Leostions.	813
Figure 3-8.	Size C Wiring Locations.	5-14
Figure 37.	Power Tenninal Strip For Model 1SU2xxx Controllers	315
Figure 3-8.	Power Terminal Strip For Model 18U4xxx and	
Careford Accession of	15U5xxx Conhollera.	3-15
Figure 3 9.	Control Term nal Strip.	3 18
Figure 3-10.	Speed Paterence Input Wiring	3-21
Figure 3-11.	Analog Meter Output Wiring.	3-21
Figure 3-12	IET/Controller Burning Wiring	8-22
Figure 3-13.	Start-Stop Control Wiring.	\$-23
Figure 3-14.	IET Reset Control Wining.	3-23
Figure 3-15.	Forward/Reverse Control Wining.	5-24
Figure 3-16.	Coast-Stop Pushbulton Control Wiring.	3-24
Figure 3 17.	J6 Jumper Sattings.	3 25
Figure 3-18.	Dynamic Braking Output Wiring for	
	Model 1SU2xx Controllers.	3-27
Figure 3-19.	Dynamic Braking Output Wiring For	
002053029.95	Models 1904xxx and 1905xxx Controllers	3-28
Figure 4 1.	SP500 Controlle: Keypad and Display.	4 1
Figure 5-1.	Factory De'ault Settings	5-1
Figure 5-2	Manual Torque Boost Adjustable Bange	510
Figure 5-3.	Volla/Hertz Curve,	5-11
Figure 7-1.	System Black Diagram.	72
Figure 7-2.	System Block Diegram (cont'c)	7-3
Figure 7-3.	D-C Bus Terminals on Model 1SU2xx Controllers	7-5
Figure 7.4.	D-C Bus Terminals on Model 1SU4xxx and	
	1SUSxxx Controllers	7-5

List of Tables

lable 1-1-	Controller Model Notation
Table 2-1	Model 1SU14001 Controller Ratings with Single-Phase Power 2-1
Ishia 2-2.	Model 1SU2xor Controller Helings with Single-Phase Prover 2-1
Table 2-3.	Model 15U2xxx Controller Ratings with Three-Phase Power 2-2
Table 2.4.	Model 1SU4xxx Controller Ratings with Three Phase Power 2.8
Table 2-5.	Model 1SU5xxx Controller Ratings with Three-Phase Power 2-4
Table 2-6.	Controller Optional Kils
Table 2-7.	Controller Power Loss
Table 3-1	Controller Model Dimensional Size
Jahle 8.2.	Single Phase A C Input Line Branch Circuit Primert on
Table S-S.	Three-Phase A-C Input Line Branch Circuit Protection for
	Model 1SU2xxx Controllers
Table S-4.	Three-Phase A-C Incut Line Branch Circuit Protection for
	Model 15U/xxx Controllera
Table 3-5.	Three Phase A-C Input Line Branch Circuit Protection for
	Model 1SU5xxx Controllers
Table 3-6.	Recommended Power Wire Size
Table 8-7.	Power Terminal Tightening Torques
Table S-6.	Recommended Control and Signal Wire Sizes
Table 3-9.	Control Terminal Strp Tightening Torques
Table 6-*	D-C Bus Voltage Value
Table 7 *.	Fault Code Action List
lable 7-2.	Heplacement Parts I Ist For Model 18UPxxx Controllers
Table 7-5.	Replacement Parts List For Model 1SU4xxx Controllers
Table 7 4.	Replacement Parts List For Model 1SU8xxx Controllers
	THE REPORT OF A DESCRIPTION OF A

1: Receive and Accept Shipment

DANGER

ONLY QUALIFIED ELECTRICAL PERSONNEL FAMILIAR WITH THE CONSTRUC-TION AND OPERATION OF THIS EQUIPMENT AND THE HAZARDS INVOLVED SMOULD INSTALL, ADJUST, OPERATE, AND/OR SERVICE THIS EQUIPMENT. READ AND UNDERSTAND THIS MANUAL IN ITS ENTIRETY BEFORE PROCEED-ING. FAILURE TO OBSERVE THIS PRECAUTION COULD RESULT IN SEVERE BODILY INJURY OR LOSS OF LIFE.

About this Instruction Manual

The SP500 A C V★S* Controller products described in the instruction manual are manufactured by Reliance ⊡eptric Industrial Company.

This instruction manual provides the information necessary to install, start up, operate and troubleshoot the SP500 controller.

Section 2 provides the specifications required to ensure properoperation of the controllers.

Section 3 instructs on how to install the controller along with, necessary wiring, and precautions.

Keypad operation is described in Section 4. A thorough understanding of the keypad, display and indicators is necessary to adjust, start-up, operate or troubleshoot the SP600 controller.

Section 5 instructs on how to acjust the drive functions and the defaults as shipped from the factory.

Section 6 gives Controller Stertup procedures.

Section 7 provides troubleehooting information and fault code conditions, replacement parts, and illustrates the drive functionality through a system functional block diagram.

Section 8 contains a Glossary of common terms used throughout this instruction manual.

Section 9 is an a phatetical index arranged by subject to be used to locate specific information quickly.

Identify the Controller

Each Reliance Electric SP500 controller can be identified by its model number (for standard controllers) or by the sales order number (for oustomer-specified controllers). This number appears on the shipping tabel and is stamped on the controller nameplate. (Refer to concoller nameplate for more information.) Verify that the model number shown on the nameplate matches the shipping label. Refer to this model number whenever discussing the eculpment with Reliance Electric personnel.



The standard model number describes the controller as follows:

Note: All SPS00 controllers described in this manual function the same. However, when necessary, ciliferences between the specific controller models will be pointed out by using the notation in Table 1.1.

Table	1.1.	Controller	Model	Notation
-------	------	------------	-------	----------

Model Notation	Input Voltage (VAC)	Horsepower (Hp)
15U2xxx	208-230	1, 2, 3, 5, 7.5, 10
1SU4xex	380-460	1, 2, 3, 5 (4kw) 7.5, 10
1SU5xxx	575	1, 2, 3, 5, 7.5, 10

Controller Description

SP500 controllers operate on a three-phase (single-phase) A-C power source at the expropriate rated input vollage. As shown in Figure 1-1, the A-C power source is supplied to the controller's input terminate and passes through three suppressors (MOV). The suppressors are used to limit voltage transients within the maximum voltage rating of the clode module.

The diode module rectifies the incoming A-C source to a constant D-C voltage. This D-C voltage is then applied to an internal power supply and to a large D-C lous capacitor. The internal power supply is a D-C to D-C type to provide the nacessary voltage and current source required within the controller. The large D-C bus capacitor filters the D C voltage which then enters an IGDT (insulated gate bipplar transistor) inverter bridge. Under software control from an internel microcontroller, the IGBT transforms the constant D-C voltage into a PWM (Pulse-width Modulated) voltage signal corresponding to the variable voltage and variable frequency source selected for the motor.

The IGBT inverter bridge is switched by the microcontroller using a 4. 6, or 8 kHz carrier frequency. A low carrier frequency will maximize the power rating of the controller but also increase acoustic noise. A higher carrier frequency selection reduces acoustic noise but results in a cereting of the controller output iPefer to Specifications, Section 2). The microcontroller also provides for operational input data from the local operator interface and operates the controller.

The microcontroller directs the overall operation of the SP500 controller for adjustable-speed performance of A-C induction and synchronous motors. User programmable functions, contigure the controller's performance to fit motor characteristics and the users end application. (Hefer to Motor Applictions in Section 2, and Adjusting Drive Functions, Section 5 for more information.) External control of the controller is provided by the local operator interface or user installed wiring.

The local operator interface contains a display, indicators, and control and programming keys. (Refer to Section 4. Keypad and Display Operation, for more information.) In addition to using the local operator interface, user writing can be installed for operational control, motor abeed acjustment, analog metering, dynamic braking (optional) and an emergency stop (customer supplied). (Refer to Installation and Wiring, Section 3).

The controller is intended to operate under trip free (fault) characteristics. The controller uses selected signals to extend the acceleration (starting) and deceleration (stopping) rates of the motor. When a fault does occur, however, the controller's microcontroller generates an instantaneous electronic trip (IET) signal to turn the controller off (coset-to-etop). An indication of the IET fault, which occured, is stored in the controller and can be displayed on the local optrator interface. After a fault, the STOP/RESET key or a winst IET RESET pishoution switch must be pressed (meeting the IET signal) to clear the fault from the controller (Refer to Section 7, Troubleehooting, Fault Codes, and Replacement Paris). Figure 1-1. Controller Functional Block Diagram



4

Controller Nameplate

The controllar nameplate is located on the base in the upper right side of the controller. The nameplate gives information such as Model Number, Controller rated single, etc. Refer to Figure 1-2.

			u /w		-
SER NO			#/N		
1000	KVA	HP@0.8P	HI/M		
AC NPUT	VOLTS	MAX	AMPS	HZ	Fr
AC OUT	VOLIS	MAX	AMPS	100.00	
SHORT CIRCI	UIT SYM RMS	RATING			

Figure 1-2. Nameplate Information.

Note: Actual nameclate configuration is subject to change.

Receive and Accept the Shipment

Reliance Flectric's terms of sale, in all instances, are FO.R. point of origin. The user is responsible for thoroughly inspecting the equipment before accepting shipment from the transportation company.

I all the items called for on the bill of lading or on the express receipt are not included or if any items are obviously damaged, do not accept the shipment until the freight or express agent makes an appropriate notation on your freight bill or express receipt. If any concession loss or damage is discovered later, notify your freight or express agent within 15 days of receipt and request that he make an inspection of the shipment. Keep the entire shipment intect it its original shipping container.

The user is responsible for making a claim against the Carrier for any shortage or camage occurring in transit. Claims for loss or damage in shipment must not be deducted from the Reliance Electric invoice, nor should payment of the myoice be withheld while awaiting adjustment of such claims alnos the Carrier guarantees safe delivery.

File a Return Request

- Io return equipment, send a written request to Reliance Hectric within ten days of receipt.
- Do not return equipment without a numbered Equipment Raturn Authorization (ERA) from Ratianoa Electric.
- Reliance Electric reserves the right to inspect the equipment on site.

Storage until Installation

After receipt inspections, repack the controller in its original shioping container until installation, if a period of storage is expected, store in the original shipping container with its internal packing.

To ensure satisfactory operation at atsrtup and to maintain warranty coverage, store the equipment:

- In its original shipping container in a clean, dry, safe place.
- within an ambient temperature range of -40°C to 85°C (-40°F to 149°F)
- within a relative numidity range of 5 to 85% without condensation.
- away from a highly corrosive atmosphere. In harsh environments, cover the shipping/storage container.

SP500 Demo Packages – Model 1SU14001 Only

CAUTION: The 1SU14001 controller has been modified to operate with a single phase, 115 VAC input. Do not operate this controller with 230 VAC input. Failure to observe this precaution could result in clamage to, or destruction of the equipment.

> The controller model number, 18014001, supplied in the SP500 demo package is specially configured to operate on 1 phase, 115 volt power and is intended for demonstration purcoses only. It will operate an unloaded motor, model number P56X3005. Refer to Table 2-1 for the controller's current and voltage data.

Note: The regulator card used in this domo is part number 0.56911-8A.

2: Specifications

Controller Specifications

SP500 Model 1SU2xxx, Model 1SU4xxx, and Model 1SU5xxx Controllers are intended to operate from a three phase. A C power source at the rated voltage listed on the controller nameplate (See Figure 1-1). Each type of controller model can operate on either a 50 Hz or 80 Hz line frequency. The controllers provide three-phase variable voltage and variable frequency to the motor. Some Mode 1SU2xxx Controllers can also operate from a single-phase voltage source. Three-phase and single-phase controller current ratings are listed in Tables 2-1 (hru 2-5.

Table 2-1. Model 15U14001 (Demo) Controller Ratinge with Single-Phase Power

Model Number	Туре	Input voltage (VAC)	Oulpui Amps (A)	Input Ampa (A)	input (KVA)
1SU14001	Demo	110-115	2.0	5.2	0.6

Model Number	Type ^(2, 3)	Input voltage (VAC)	Output ⁽²⁾ Amps (A)	Input Amps (A)	inpul (KVA)
18021001 18024001	1Ho NEMA 1 1Ho NEMA 4X/12 4010	200-230	2	5. D	
	4 kHz Carrier		1.7	5.0	1.3
	6 kHz Carrier		1.7	5.0	1.3
a utore to constant	8 kHz Carrier		1.7	ð.0	1.3
1SU21002	2Ho NEMA 1 40°C	200-230			
	4 kHz Carrier		7.5	19.1	4.4
	6 kHz Carrier		7.0	17.2	4.0
	8 kHz Carrier		6.5	15.3	3.5
	2Ho NEMA 1 3510	200-230			
	4 kHz Carrier		7.5	19.1	4.4
	6 kHz Carrier		1.2	19.1	4.4
	8 kHz Carrier		7.0	17.2	4.0

Table 2-2. Model 1SU2xxx Controller Ratings with Single-Phase Power⁽¹⁾

Note: (1) Single phase input power applies only to model 1SU2xxx controllers.

> (2) To size the controller for Motor Namepiale Horsepower and Motor Namepiate Amperes, refer to Motor Applications in this section.

(3) NEMA 4X is indeer only.

Model		Input voltage	Output ⁽¹⁾ Amps	Input Amps	Input
Number	Type ^(1, 2)	(VAČ)	(A)	(A)	KVA
1SU21001	1Hp NEMA 1	200-230			
	40°°C				
	4 kHz Carrier		5.0	7.0	2.6
	β kH∠ Carrier		5.0	7.0	2.6
	8 kHz Carrier		5.0	7.0	2.B
1SU24001	* Hp NEMA 4X/12	203-230			
	43°C				
	4 kHz Car1er		4.5	6.4	2.6
	6 kHz Car1er		3.6	5.2	2.0
	B <hz car1er<="" td=""><td></td><td>3.6</td><td>5.2</td><td>2.0</td></hz>		3.6	5.2	2.0
	1 Hp NEMA 4X/12	203-230			
	35"C				
	4 kHz Carrier		4.5	6.4	2.6
	5 kHz Carter		2.0	5.8	2.3
	BikHz Carton		2.0	5.8	2.3
1SJ21002	2Hp NEMA 1	203 230			
15U24002	2Hp N⊟MA 4x/12				
	40°C				
	4 kHz Carrier		7.5	9.9	4.0
	5 kHz Carter		7.0	9.3	3.6
	B Kilz Carrier		6.5	8./	3.5
1SU21002	2Hp NEMA 1	200-230			
	85°G				10
	4 KHZ Garner		7.5	9.2	4.0
	BikHz Garner		7.9	9.2	4.U 2.0
40104000	B KHZ Garlor		7.0	9.5	<i>a.</i> b
15021003	AHD NEMA 1	200-230			
15024003	300 NEMA 42012				
	40°C		10.2	12.5	5.0
	4 KHZ Garler 6 xHz Carler		9.5	11 3	4.5
	B (Hz Carler		85	10.0	4.0
18.121025	54n2 Gdridi 5Hn NEMA 1	900 930	0.0	10.0	1.9
18024005	SHO NEMA 4X/12	101 100			
1000	40"C				
	4 kHz Carrier		14.2	17.2	6.9
	6 kHz Carrier		12.8	15.5	6.2
	8 dlz Carior		11.4	13.6	5.Б

Table 2-3. Model 1SU2xxx Controller Ratings with Three-Phase Power

Note: (1) To size the controller for Motor Nameplate Horsepower and Motor Nameplate Amparee, refer to Motor Applications in this section.

(2) NEMA 4X is incort only.

Model Number	Tupe(1, 2)	Input voltage (VAC)	Output ⁽¹⁾ Amps (A)	Input Amps (A)	Input (KVA)
1SU41001 1SU44001	1Hp NEMA 1 1Hp NEMA 4X/12	380-460			
	40°C 4 kHz Carrier 6 kHz Carrier 8 kHz Carrier		2.1 1.9 1.7	2.5 2.3 2.0	2.0 1.8 1.6
1SU41002 1SU44002	2H2 NEMA 1 2Hp NEMA 4X/12 40°C	380-460	1.1	2.0	1.0
	4 kHz Carrier 6 kHz Carrier 8 kHz Carrier		3.4 3.1 2.8	4.2 3.8 3.4	3.3 3.0 2.7
1SU41000 1SU44003	3He NEMA 1 3He NEMA 4X/12 40 °C	380-460			
	4 kHz Carrier 6 kHz Carrier 8 kHz Carrier		5.3 4.8 4.3	6.4 5.8 5.2	5.1 4.6 4.1
1SU41005 1SU44006	5H¢ NEMÁ 1 oH¢ NEMA 4X/12 40°C	400-460			
	4 kHz Carrier 6 kHz Carrier 8 kHz Carrier		9.2 7.4 6.6	9.9 8.9 8.0	8.0 7.2 6.4
1SU41005	4 KW NEMA 1 40°C 4 Mile Castler	380	0.7	10.5	70
	6 kHz Carrier 8 kHz Carrier 8 kHz Carrier		5.7 7.8 5.9	9.4 8.4	8.3 5.6
1SU41007 1SU44007	7.5Hp NEMA 1 7.5Hp NEMA 4X/12 40°C	380-460			
	4 kHz Carrier 6 kHz Carrier 8 kHz Carrier		11.1 10.0 8.9	13.4 12.1 10.8	10 7 9.7 8.6
1SU41010 1SU44010	10Hp NEMA 1 10Hp NEMA 4X/12 4027	380-460			
	4 kHz Carrier 6 kl Iz Carrier 8 kHz Carrier		14.2 12.8 11.4	17.2 15.5 13.8	18 7 12 4 11.0

Table 2-4. Model 15U4xxx Controller Ratings with Three-Phase Power

Note: (1) To size the controller for Motor Namepiale Horaepower and Motor Namepiate Amperes, refer to Motor Applications in this section.

(2) NEMA 4X is indeer only.

Model	n = -(1, 2)	input voltage	Output ⁽¹⁾ Ampe	Input Ampe	Input
Number	Type: 5-5	(VAC)	(A)	(A)	(RVA)
12051001	1Hp NEWA 1	5/5			
18054001	1Hp NEWA 4X/12				
	40°C				
	4 kHz Cattier		1.6	20	2.0
	6 kHz Carrier		1.5	18	1.8
	8 kHz Carrier		1.3	16	1.6
19U\$1002	2Hp NEVA 1	575			
ISU54002	2Hp NEWA 4X/12				
	40°C				
	4 kHz Gartier		2.7	84	3.3
	6 kHz Gartier		2.5	31	3.0
	8 kHz Carrier		2.2	28	2.7
15U51003	SHp NEMA 1	575			
15U54003	3Hp NEWA 4X/12				
	40%)				
	4 kHz Gartier		4.3	52	5.1
	6 kHz Gaitiər		3.9	47	4.6
	8 kHz Gartier		a.5	42	4.1
15U51005	5Hp NEMA 1	575			
12U54005	5Hp NEMA 4X/12				
	40°C				
	4 kHz Cartler		6.2	75	7.5
	∂ kHz Caπler		5.6	68	8.8
	a kHz Cartier		5.0	60	6.0
1SU51007	7.5Hp NEMA 1	575			
1SU54007	7.5H¢ NEMA				
	4X/12				
	40°C				
	4 kHz Carrier		9.0	10.9	10. 9
	6 kH∠ Carrier		8.1	вa	B.9
	8 KHz Cartler		7.2	87	6.7
1SU51010	10Hp NEMA 1	575			
1SU54010	101 lp NEMA				
	4X/12				
	40%				
	4 kHz Carrier		12.0	14.5	14.4
	∂ kHz Caπler		10.8	13.1	18.0
	8 kHz Caπler		9.6	11.6	11.5

Table 2-5. Model 1SUSxxx Controller Ratings with Three-Phase Power

Note: (1) To size the controller for Motor Nameplate Horsepower and Motor Nameplate Ampares, refer to Motor Applications in this section.

(2) NEMA 4X is incort only.

Service Conditions

٠	Ambient temperature:	0 'C to 40''C (32 'F to 104' F) for enclosed controllers
	Storage temperature:	-40"C to 65"C (-40°F to 149°F)
ě	Atmosphere:	5 to 95% non-concensing reletive humidity
•	Elevation:	lo \$300 feet (1000 meters) above see level wilhout derating.
		For every 300 feet (91.4 meters) from 3300 - to 10,000 feet (1001 to 3033 meters), perate the current by 1%
		Above 10,000 feet (3033 meters) – Consult your Reliance Electric Sales Office
٠	Line frequency:	50+6% Hz and 60+6% Hz
٠	Line vollage variation:	±10%
•	A-C Line distribution capacity (maximum):	Model 1SU2xxx Controllers 100KW for 230 WAC, three phase with a maximum of 5,000 symmetrical amp fault current capacity.
		Model 1804xxx and 1805xxx Controllers 1000KVA for 460 and 575 VAC, three-phase with a maximum of 25,000 symmetrical amp fault current capacity.

Controller Input/output Specifications

The controller incut/output apecifications are subject to some of the 17 functione (19 function in total) which can be acjusted for user specified applications. (Refer to Section 16, Adjusting the Drive Functions)

Controller Inputs

- Speed Reference input: 5KQ Potentiometer (3 10 VDC) or 0 20 mA current-loop (Jumper selectable, Refer to Section 3)
- Note: The controller provides +15 VDC buffered through a 1.875 KΩ resistor, a 500Q current loop resistor and an associated reference.

Control logic

The controller uses an internal 24 VDC power supply to provide the required voltage for control signals. Enabling or disabling a control signal requires that a contact (switch) be opened or closed.

Stop: Open Cornart

(Contact must be closed when drive is running. An open contact turns the controller "off". The drive will remain and/or be held off as long as contact is open.)

 Start: Open to Closed Contact Transition – momentary or fixed contact obsure

(Ecge-sensitive control input signal which must see an open to closed transition.)

 ET Reset: Open to Closed Contact Transition – momentary or fixed contact closure

> (Edge-sensitive control input signal which must see an open to closed transition.)

Forward/

Reverse: Open Contact – asserting the Forward Direction –or– Closed Costs d

Cinseri Contect – asserting the Beverse -Direction

Function Loss:

Open Contact

(Contact must be closed when drive is running. An open contact turns the controller "off". The citive will remain and/or be held off as long as contact is open.)

Controller Outputs

Analog Output, metering:	0 to 10 VDC acaled signal (The signal represents the same display mode currently shown on the local display of the controller.) The display mode can be one of the following:
Output Volts	 (0 - 255 VAC, for Model (SU2xx) Controllers)
	(0 505 VAC, for Model 1SU4xxx Controllers)
	(0 – 632 VAC, for Model 1905xxx Controllers)
% Load (AMPS)	 0 – 200% (Percentage of output amps based on controller nameplate)
BPM/	
Engineering Unit	 Minimum to Meetmum HPM - or
	Minimum to Meximum of any
	Engineering unit
	(user application dependent – Refer to Function F 08 in Section 5)
% External Speed	
Reterence	 0 – 100% (Percentage of the external reference signal range)
let Changing the mod will change the sig selector mode sig be the same mode Up.	e displayed on the local operator interlace na on the meter analog output. The played when the drive is powered down will displayed when the controller is powered
	Analog Outout, metering: Output Volts % Load (AMPS) RPM/ Engineering Unit % Exernal Speed Reterance let Changing the mode will change the sig selected mode aisp be the same mode up.

- Dynamic Braking Signal: Dynamic braking control signal used by the controllara optional Dynamic Braking Kil
- ICT/ Controller Running: 116 VAC/ 24 VDC, 1/2 Amp relay output (1 Form A and 1 Form B contact wired with a angle common)

Controller Application Data

 Displacement Power Factor.

0.96

- Maximum Load:
- Overcurrent Tric (ET):
- Linearity:

 55% for one minute (based on controller nameplate rating)

203% rated drive current

+0.05% (Speed reference to output frequency)

DANGER

THE SP500 CONTROLLER IS INTENDED TO OPERATE THE MOTOR AT A PREDETERMINED MINIMUM SPEED UNLESS DISCONNECTED FROM THE POWER SOURCE. TO PREVENT INADVERTENT CONTACT WITH OPERAT-ING EQUIPMENT, THE USER MUST VERIFY THAT THE MOTOR OUTPUT SHAFT WILL ROTATE AT ALL COMBINATIONS OF LOAD AND OUTPUT SPEED REQUIRED BY THE APPLICATION. FAILURE TO OBSERVE THIS PRE-CAUTION MAY RESULT IN SEVERE BODILY INJURY OR LOSS OF LIFE.

- Minimum Frequency: 3 to 30 Hz
- Maximum Frequency: 30 to 240 Hz
- Long Term Frequency Stability: 0.01%
- Line Dip Ride Through Capacity: 500ms.

Motor Applications

To obtain Motor Nameplate Horsepower, the controller's (alre wave) output ampere rating, at the carrier frequency selected, should be equal to or greater than the motor nameplate current. If the Motor Nameplate Amperes are HIGHER than the controlkers (sine wave) output ampere rating, the motor HORSEPOWER should be DERATED by the ratio of controller (sine wave) output ampere rating (at the selected carrier) to the motor nameplate current.

Single-Motor Applications

The controller and motor must be sized for the load and speed requirements of the specific application.

I the motor is overaized the motor operating current must not exceed the chice's rated output current (at a selected carrier frequency). In addition, the motor horsecower must not be more than one size larger than the controllens horsepower rating.

If the motor will be operated BELOW one half of the motor's rated speed, the motor overload relay may not protect the motor because of reduced cooling action due to the reduced speed. A motor themostal, internal to the motor, should be installed because it monitors the actual temperature of the windings.

Mulli-Motor Applications

One controller can run two or more motors. Adhere to the following requirements in assure correct drive operation:

 When starting and stopping all the motors at the same time (using the drive for starting and stopping), the sum of the full-bac sine wave currents of all the motors must be equal to on ess than the maximum sine wave output current, at the carrier frequency selected, for the drive.

For example:

IFLA IPLA + IPLA = IPLA (MOTOR 2) (MOTOR 3) (TOTAL LOAD) (MOTOR 1: Where:

 $I_{TLA} \lesssim 100\%$ rated drive subput at the carrier frequency selected

- When one or more of the motors connected to the output of the controller are to start independently (using a secondary switching device to add or remove the motor from the orive);
 - Any motor that starts or stops, while the controller is running, must have a current rating less than 10% of the maximum sine wave current rating of the crive at the selected carrier frequency.
 - The sum of the maximum full load sine wave currents of all the inotors connected continuously to the controller must be less than the maximum drive output current rating under any conditions.
- Note: Each motor requires separate overload protection (i.e. a motor relay or a motor thermostal).

Motor Lead Lengths

For applications using one inclor connected to the controller, individual motor lead lengths cannot exceed 250 feet per phase.

For applications where multiple motors are used, total lead lengths on each phase cannot exceed 250 leet, and each motor connection cannot exceed 250 feet. For example, Figure 2-1 illustrates correct application connections.



Figure 2-1. Motor Lead Length Connections.

If total lead length exceeds 250 feet, nuisance tripping may occur. These tripe are caused by capacitive current flow to ground and are not an indiction of any problem with the controller. If the lead, ength must be exceeded, output line reactors or other steps must be taken to correct the problem.

Controller Optional Kits

An optional braking kit is available for each control or as shown in Table 2-8.

Table 2-8. Controller O	plional Kita
-------------------------	--------------

Description	Model Number	Instruction Sheet
Wa	ue 1SJ2xxx Controllers	
Dynamic Braking Kit	2DB2005 (UU/CSA)	D2-S178
Ma	oe 18.J4xxx Controllers	·
Dynamic Braking Kit ⁽¹⁾	2DB401010U/CSAI	D2-5179
Ma	ue 1SJ6xxx Controllers	e svincorie
Dynamic Braking Kit ⁽¹⁾	2D8501010U/CSAL	D2-S160

Notes: (1) Dynamic Breking Kit for model 1SU4xxx (460VAC) and Model 1SU5xxx (575VAC) Controllers require connections to the DB 10V Power Supply (Pefer to Figure 3.8.)

Controller Power Loss

The typical full load power loss watte under all operating carrier frequencies is shown in Table 2-7.

Horsepower (Hp)	Input Voltage (YAC)	Typical Full Load Power Loss (WATTS)
9000000	Model 15U2xxx Control	lers
1/4-1	¥	70
2		120
3	200-230	210
5		250
	Model 1SU4xxx Control	lens
1/4-1		60
2	1	100
3	5-9-9-010A804	140
5 (4 kw)	400440 (380)	160
7.5	1	210
10	Martin Constants and the second	250
	Model 1SU5xxx Control	lers
1/4-1		50
2	1	90
3		120
5	ā75	150
7.5	1	180
10	1	220

Table 2-7. Controller Power Loss (Waits)

3: Installation and Wiring

DANGER

ONLY QUALIFIED ELECTRICAL PERSONAL FAMILIAR WITH THE CONSTRUCTION AND OPERATION OF THIS EQUIPMENT AND THE HAZ-ARDS INVOLVED SHOULD INSTALL, ADJUST, OPERATE, AND/OR SERVICE THIS EQUIPMENT. READ AND UNDERSTAND THIS MANUAL IN ITS EN-TIRETY BEFORE PROCEEDING. FAILURE TO DESERVE THIS PRECAUTION COULD RESULT IN SEVERE BODILY INJURY OR LOSS OF LIFE.

DANGER

THE USER IS RESPONSIBLE FOR CONFORMING TO THE NATIONAL ELEC-TRICAL CODE (NEC) AND ALL OTHER APPLICABLE LOCAL CODES. WIR-ING, GROUNDING, DISCONNECTS, AND OVERCURRENT PROTECTION ARE OF PARTICULAR IMPORTANCE. FAILURE TO OBSERVE THIS PRECAU-TION COULD RESULT IN SEVERE BODILY INJURY OR LOSS OF LIFE.

CAUTION: Use of power correction capacitors on the output of the controller can result in terratic operation of the motor, nuisar ce tripping, and/or permanent dam age to the controller. Remove power factor capacitors before proceeding. Failure to observe this precaution could result in damage to, or destruction of the equipment.

Planning and Location

Planning before installation is necessary to ensure that the controller environment and operation concitions are satisfactory. Reed and follow the recommendations advised in this section before proceeding with the installation.

- 1. Verify that the controller can be kept clean and cool.
- Check that the controller will be away from oil, coplants, or other airborne contaminants.
- Check that the temperatures within the vicinity of the controller are between 0° to 49°C (32° to 104°F).
- Check that the relative humicity is between 5 and 95% noncondensing.
- Do not install above 3300 feet (1000 meters) without derating. For every 300 feet (91.4 meters) above 5300 feet, derate the current rating by 1%. Consult Heliance Electric Sales for operation above 10,000 feet.
- Check that the area chosen will allow the space required for sir flow, around the controller.

Controller Mounting

The dimensional size of SP500 Controller models will vary depending on horsepower (HP) rating. Heler to Table S-1 to determine the appropriate dimensional alze, (given in type A,B, or C) for model 1SU2xxx, 1SU4xxx, and 1SU5xxx controllers. When mounting a controller, adhere to the guidelines and physical diagrams for the specific size type (A,B, or C).

Horsepower (HP)	Dimensional Size (Type)	Controller Model Number(s)
N	Iodel 1SU2xxx Controller	15 (230VAC)
Deniu	đ	1SU14001
4	A	1SU21001, 1SU24001
2		15U21002
2		1SU24002
3	C	15U21003, 15U24003
5	82.	1SU21005, 1SU24005
M	odel 15U4xxx Controller	s (460 VAC)
1		1SU41001, 1SU44001
2	в	15041002, 15044002
3		1SU41003, 1SU4400R
5 (4kw)		15041005, 15044005
7.5	Se la companya de la	15U41007, 15U44007
10		1SU41010, 1SU44010
М	odel 15U5xxx Controller	a (575 VAC)
1	6	1SU51001, 1SU54001
2	36	1SU51002, 1SU54002
3	8	15051003. 15054008
5		1SU51005, 1SU54005
7.5	-	1SU51007, 1SU54007
10	G	1SU51010, 1SU54010

Table 3-1. Controller Model Dimensional Size

Size A Controller Mounting (230VAC 1&2 HP)

- Loosen the four (4) attaching screwe and remove the cover from the controller.
- In the location selected, mount the controller vertically using the two (2) mounting holes provided in the controller base. See Figure 3 1.

- Use the to lowing as reference to provide adequate clearances for air ventilation;
 - At least 2 inches from the sides and 4 inches from the top and bottom of the controller to adjacent non-heat producing equipment, such as a cabinet wall.
 - At least 2 inches from the sides and 10 inches from the top and bottom of adjacent controllers. For best air movement with three or more controllers, do not mount the controllers in a vertical stack (i.e. offset (stagger) the controllers).



Figure 3-1. Size A Physical Dimensions.

Size B Controller Mounting (460 VAC 1-5 HP & 575 VAC 1-5 HP)

- In the location selected, mount the controller vertically using the four (4) mounting holes provided on the controller base. See Figure 3-2.
- Use the following as reference to provide adequate clearances for air ventilation;
 - At least 4 inches from the sides and 4 inches from the top and bottom of the controller to adjacent non-heat producing equipment, such as a cabinet wall.
 - At least 4 inches from the sides and 10 inches from the top and bottom of adjacent controllers. For best air movement with three or more controllers, coinct mount the controllers in a vertical stack (i.e. offset (stagger) the controllers).



Figure 3-2. Size 8 Physical Dimensione.

Size C Controller Mounting (230 VAC 2 (NEMA 4), 3 and 5 HP, 460 VAC 7.5 and 10 HP, and 575 VAC 7.5 and 10 HP)

- In the location selected, mount the controller vertically using the four (4) mounting holes provided in the controller base. See Figure 3-3.
- Use the following as reference to provide adequate clearances for air ventilation:
 - At least 4 inches from the sides and 4 inches from the top and bottom of the controller to adjacent non-heat producing equipment, such as a cabinet wall.
 - At east 4 inches from the sides and 10 inches from the top and bottom of adjacent controllers. For best sin movement with three or more controllers, do not mount the controllers in a vertical stack (i.e. offset (stagger) the controllers).



Figure 3-3. Size C Physical Dimensions.

DANGER

THE NEC/CEC REQUIRES THAT UPSTREAM BRANCH PROTECTION BE PROVIDED TO PROTECT INPUT POWER WIRING. INSTALL AND DO NOT EX-CEED THE RECOMMENDED BRANCH PROTECTION RATING. FAILURE TO OBSERVE THIS PRECAUTION COULD RESULT IN SEVERE BODILY INJURY OR LOSS OF LIFE.

- Install an input disconnect in the incoming power line according to the NEC/CEC guidelines.
- Size the disconnect according to the inrush current as well as any additional loads the disconnect may supply.
- Note: Coordinate the trip rating for the in-rosh current (10–12 times fall load current; with that of a transformer (flueco). See Transformer Installation (if needed). later in this section.

DANGER

THE NEC/CEC REQUIRES THAT UPSTREAM BRANCH PROTECTION BE PROVIDED TO PROTECT INPUT POWER WIRING. INSTALL AND DO NOT EX-CEED RECOMMENDED THE BRANCH PROTECTION RATING. FAILURE TO OBSERVE THIS PRECAUTION COULD RESULT IN SEVERE BODILY INJURY OR LOSS OF LIFE.

CAUTION: The input fuse ratings listed in tables 3.2 through 3.5 are applicable for one driver per branch circuit. No other load can be applied to that fused branch circuit. Failure to observe this precaution could result in damage to, or destruction of the equipment.

- 1. Install user-supplied branch circuit protection.
- According to the values given in Tables 3-2 through 3-6, size the branch circuit protection for the specific controller model.

Model Number (1SUxxxx)	Horsepower (Hp)	Single-phase Input Voltage (VAC)	input A-C Fuse Rating ⁽¹⁾ (Amps)
1\$U14001	Вета	115	12
1SU21001 1SU24001	1	200-230	10
1SU21002	2	200-230	30

Table 3-2. Single-Phase A-C input Line Branch Circuit Protection.

Note: (1) The recommended fuse type is UL Class J, 600V, time delay.

Table 3-3. Three-Phase A-C input Line Branch Circuit Protection for Model 1SU2xxx Controllers.

Model Number (1SU2xxxx)	Horsepower (Hp)	Single-phase Input Voltage (VAC)	Input A-C Fuse Rating ⁽¹⁾ (Amps)
1\$U21001 1\$U24001	1		12
1SU21002	2		20
1SU21003 1SU24003	3	200-230	25
1SU21005 1SU24005	5		35



Model Number (1SU4киях)	Horsepower (Hp)	Three-phase Input Yoltage (YAC)	Input A-C Fuse Rating ⁽¹⁾ (Amps)
1SU#1001 1SU#4001	1	400460	Ġ
1SU41082 1SU44002	2		Б
19041008 19044003	а		12
1SU41005 1SU/4005	u.		25
1SU41005	(4Kw)	(380)	25
15U/1007 15U/4007	7.5	400-460	25
15U/ 1010 15U/ 4010	10		35

Table 3-4. Three-Phase A-C input Line Branch Circuit Protection for Model 1SU4xxx Controllers.

Note: (1) The recommended fuse type is U_ Class J, 600V, lime-delay.

Table 3-5.	Three-Phase A-C Input Line Branch Circuit Protection for
	Model 1SUSect Controllers.

Madel Number (15U5xxx)	Horeepower (Hp)	Three-phase Input Voltage {VAC}	Input A-C Fuse Rating ⁽¹⁾ (Amps)
1SU51001 1SU54001	, st	6.32	4
15U51002 1SU54002	2	ara	7
15U51003 1SU54003	3		10
15U51005 1SU54005	5		15
15U51007 1SU54007	7.5		20
15U51010 1SU54010	10		25

Note: (1) The recommended fuse type is U_ Class J, 600V, lime-delay.

Transformer Installation (If Needed)

Transformers step up or step down the voltage and can be either autotransformers or isolation transformers, isolation transformers help eliminate:

- Damaging A C line voltage transients from reaching the controller.
- Line noise from the controller back to the incoming power.
- Damaging currents, which could develop if a point inside the controller becomes grounded.

Input Transformers

f an input transformer is installed ahead of the controller, adhere to the following:

- A power disconnecting device must be installed between the power line and the primary of the transformet.
- If the power disconnecting device is a circuit breaker, the circuit breaker trip rating must be coordinated with the innush of current (10 to 12 times full load current) of the transformer.
- An input transformer rated more than 100 KVA for 230VAC (1000 KVA for 460 and 575 VAC) with less than 5% impedance should NOT be used directly ahead of the controller without additional impedance between the controller and the transformer.

CAUTION: Distribution system capacity above the maximum recommended system KVA (100KVA for 230 VAC; 1000KVA for 460 and 575 VAC; requires using an isolation transformer, a line reactor, or other means of adding similar impedance. Failure to observe these precautions could result in damage to, or destruction of, the equipment.

CAUTION: When the A-C line is shared cirectly with other SCR rectified drives, a. line reactor or optional DR kit may be required to alleviate excess D-C bus voltage. Failure to observe these precautions could result in damage to, or destruction of, the equipment.

Output Transformers

in applications requiring the use of an output transformer on the controller, contact your Reliance Electric Sales Office for assistance.

DANGER

THE USER IS RESPONSIBLE FOR CONFORMING TO THE NEC/CEC AND ALL OTHER APPLICABLE CODES. WIRING, GROUNDING, DISCONNECTS, AND OVERCURRENT PROTECTION ARE OF PARTICULAR IMPORTANCE. FAILURE TO OBSERVE THIS PRECAUTION COULD RESULT IN SEVERE BODILY INJURY OR LOSS OF LIFE.

- Remove the four (4) captive screws and remove the cover. (When installing the cover on the controller, refer to Cover Installation for NEMA 4X (Indoor only)/12 Controllers. Section 6.)
- Note: To conform to CSA requirements, when adding more than one grounding conductor wire to a single chassis ground, twist the conductors together.
 - Bun a suitable equipment grounding conductor unbroken from the controller ground terminal (See Figures 8.4, 3.5, and 3.6) to an earth ground conductor. See Table 3-6 for recommended wire sizes.
 - Connect a suitable equipment grounding conductor to the motor frame, the remote control station (if used), and the transformer. Run each conductor unbroken to the safth ground.

Motor Preparation

- 1. Install the motor according to the motor instruction manual.
- Verily that the motor is the appropriate size to use with the controller. (Refer to Motor Applications, Section 2).
- In single motor applications, verify that the total lead length on each phase does not exceed 250 feet per phase.
 -pr.

In multi-motor applications, varify that the total lead length on each phase does not exceed 250 feet, and each motor connection cannot exceed 250 feet.

Refer to Motor Applications. Section 2, for more information.

- Verily that the motor is properly aligned with the driven machine to minimize unnecessary motor loading from shall misalignment.
- If the motor is accessible while running, install a protective guard around all exposed rotating parts.

In applications requiring the use of an output fransformer on the controller, contact your Reliance Electric Sales Office for assistance.

Controller Wiring

Size and install all wiring in conformance with the NEC/CEC and all other applicable local codes. If not almostly cone, locsen the four (4) captive screws on the cover and remove the cover from the controller base. Refer to Figures 3-4, 3-5, and 3-6 for jumper and wiring locatione on size A,B, and C controllers. Follow recommended wires and tightening torques.

CAUTION: Do not route signal and control wiring with power wiring in the same conduit. This may cause interference with controller operation. Failure to observe this precaution could result in damage to, or destruction of the equipment.

> Note: As a general rule, route the signal and control witing in secerate conduits to prevent interference with controller operation.

WARNING

THE CONTROLLER IS NOT EQUIPPED WITH A COAST-STOP PUSHBUTTON. THE USER MUST INSTALL & HARDWIRED. OPERATOR-ACCESSIBLE PUSH-BUTTON THAT PROVIDES & POSITIVE INTERRUPT AND SHUTS DOWN THE DRIVE. (USE TERMINALS 10 AND 11. SEE FIGURE 3-4, 2-5, 3-5 AND 3-16.) FAILURE TO OBSERVE THIS PRECAUTION COULD RESULT IN BODILY INJURY.



Rigure 3-4. Size A Wiring Locations.






-subjectory Euplin C esis .a.c englis

Power Wiring

Size and install all wiring in conformance with the NEC/CEC and all other applicable local codes. Refer to Figures 3-7 and 3-8 when making wire connections to the power terminal step. See Table 3-6 for recommended wires sizes and Table 3-7 for power terminal tightening torque.

- Verify that the input power to the controller corresponds to the controller nameplate voltage and frequency and that the plant supply is of sufficient capacity to support the input current requirements. (Refer to Specifications, Section 2.)
- Provide a transformer between the plant power supply and the controller if the correct input line voltage is not available. (Balar to Transformer Installation in this section.)
- Size upstream branch circuit protection (fuses) according to Tables 3-2 through 3-5.



Figure 3-7. Power Terminal Strip For Model 19U2xxx Controllere.



Figure 3-8. Power Terminal Strip For Model 1SU4xxx and 1SU5xxx Controllers.

 Refer to Table 3-8 and size the input and output power wiring to handle the rated maximum controller current. (Refer to Tables 2-1 through 2-5 for maximum controller currents.)

Wiring	Terminal ⁽³⁾	Terminal ⁽⁴⁾	Wire Stzg(5)
Incut power	R, S, T	FI(L), S(Lg(, T(Lg)	
Output prover	u, v, w	$U(T_1), V(T_2), W(T_3)$	14 AWG
D-C BLB	+	-,-	-10-
DB Power		10 VDC, 10 COM	(See Note 5)
GND Terminal	GND Stud	GND Stud	1

Table 3-6. Recommended Power Wire Size (1.2)

Notes: 1) The user is responsible for the following NEC/CEC and all applicable local codes with respect to wire sizes used with the controllers. Acceptable wire sizes are shown in the Table.

- Use only copper wire with a minimum temperature rating or 60/75°C.
- 3; Terminals on Model 1SU2xx Controllers.
- 4) Terminals on Model 1SU4xxx and 1SU5xxx Controllers.
- The following is a list of the controller models where 12 AWG wire is recommanded, all other models can use 14 AWC wire:
 - 1SU21002 (Single-chaze) 1SU21005, 1SU24005, 1SU41010, 1SU44010
- Use the appropriate terminal torque as listed in Table 3-7 for wire connections to the power terminal.

Table 3-7. Power Terminal T	ightening	Torques	(In-Ibs)
-----------------------------	-----------	---------	----------

Model 1SU2xxx Terminals	Model 1SU4xxx and Model 1SU5xxx Terminals	Torque
FL, S, T,	$B(L_1), S(L_2), T(L_3),$	100
U, V, W	U(T ₁), V(T ₂), W(T ₃)	9 mir. –
-, +,	-, +, +10 VDC, 10 DOM,	12 max. In Ibs
GND Stud	GND Stud	

 The recommended routing for power wing through the controller base is as follows:

(Refer to Figures 3-4, 3-5, and 3-6.)

Without the Dynamic Braking Option -

- Size A Route only AC input leads through the bottom left opening of the controller base.
- Size B Koute A-C input leads through the bottom center (or right) opening of the controller base.
- Size C Route A-C input leads through the third opening (from the left) of the controller base.

With the Dynamic Braking oction -

Size A.B.C Route A-C input power and motor leads through the bottom right (far right) opening of the controller base as indicated in Figures 3-4, 3-5, and 3-6.

 Wire the A G input power leads as follows: (Refer to Figures 3.7 and 3-8.)

For Model 1SU2xxx Controllers - To terminals R,S, and T

For Model 1SU4xxx and 1SU5xxx Controllers – Ic terminals – $R(L_1),\,S(L_2),\,anc,T(L_3)$

- (Note: For single-phase units, connect the input power leads to any one combination of R&S or S&T.)
- Wire the motor leace as follows: (Refer to Figures 3-7 and 3-8.) For Model 1SU2xxx Controllers – To terminals U.V. and W

For Model 1SU4xxx and 1SU5xxx Controllers – To terminals $U(T_1), V(T_2), and W(T_3)$

Control and Signal Wiring

Size and install all wiring in conformance with the NEC and all other applicable local codes. Refer to Figure 3-9 when making wire connections to the control terminal strip. See Table 3-8 for recommended wires sizes. Refer to Table 3-9 for terminal lightening torque.

CAUTION: Do not route signal wiring with power wiring in the same conduit. This may cause interference with controller operation. Failure to observe this precaution could result in damage to, or destruction of the equipment.



- 1. For all signe wiring use twisted pair wire.
- For distances of up to 1000 feet, use a minimum of #22 AWG whe.
- 3. Refer to Table 3-8 and a ze control and signal wiring.

Wiring	Terminal	Terminal Wire Size	
Speed Reterence	1, 2, 3		
Analog output	4, b		
Step	6.11		
Start	7,11		
IET Reset	8.11		
Forward/ Peverse	8,11 22-	22-14 AWG	
Function Loss	10, 11		
DB Control	12, 13		
IFT: Challen	14, 16		
Bunning ⁽¹⁾	-01-		
including ()	14, 16	-	

Table 3-8. Recommended Control and Signal Wire Sizes

- Notes: 1) Terminals 14 and 15 provide a N.O. relay contact while, terminals 14 and 16 provide a N.O. relay contact. (Function of contacts controlled by setting of Function F-09. Refer to Adjusting the Drive Functions, Section 5.)
 - Use the appropriate terminal torque as listed in Table 3-9 for control and signal wire connections to the control Terminal Strip.

Table 3-9. Control Terminal Strip Tightening Torques (in - ibs)

Torque	
7 in - Ibs max.	

 Route the signal wiring (External Space Reference, Analog Output, and IET/ Controller Running) and the control wiring (Stop, Start, IET Reset, Forward/Reverse, Function Loss) to the controller as follows:

Refer to Figures 3 4, 3 5, and 3 6.

S ze A - Route the wiring as follows:

- All signal wiring through the top left opening in the control er.
- All control wing through the top right opening in the controller.
- Size B Route the wiring as follows:

Without the Dynamic Braking Option -

- Route all signal wring through the bottom laft opening of the controller, around the bracket assembly and through the wiring support dip.
- Route all control wiring through the bottom center opening of the controllet around the bracket assembly and through the wiring support clp.

With the Dynamic Braking Option -

 Route all signal and control wiring through the bottom laft opening of the controller, around the bracket assembly and through the wring support clip.

Size C - Route the wiring as follows:

- Route all signal wiring through the brittom loft opening of the controller, around the bracket assembly and through the wiring support dip.
- Boule all control wining through the second opening (from the left) on the controller, around the bracket assembly and through the wining support clip.
- Wire the signal leads. Refer to Figures 3-10, 3-11, and 3-12 for External Speed Reference, Analog Output, and IET/ Controller Bunning connections.



Figure 3-10. Speed Reference Input Wiring



Figure 3-11. Analog Meter Output Wiring.



Figure 3-12, IET/Compolier Output Winng.

 Wire the control leads. Refer to Figures 3-13, 3-14, 3-15, and 3-16 for Stop, Start. IET Reset, Forward/Reverse, and Function Loss connections.

Note: Remove the factory-installed jumper at terminals 10 and 11 (Refer to Figure 3 18) to install a normally closed, membershall amergency stop pushbutton as shown in Figure 3-6.



Figure 3-13, Start-Stop Control Wiring.



Figure 3-14. IET Reset Control Wiring.







THE CONTROLLER IS NOT EQUIPPED WITH A COAST-STOP PUSHBUTTON. THE USER MUST INSTALL A HARDWIRED, OPERATOR-ACCESSIBLE PUSH-BUTTON THAT PROVIDES A POSITIVE INTERRUPT AND SHUTS DOWN THE DRIVE. (USE TERMINALS 10 AND 11. SEE FIGURE 3-16-) FAILURE TO OB-SERVE THIS PRECAUTION COULD RESULT IN BODILY INJURY.



Figure 3-16. Coast-Stop Pushbutton Control Wiring.

Remote Analog Input Reference Jumper Setting

Located on the regulator coard (Refer to Figures 3-4, 3-5 and 3-8) is a remote speed reference input jumper (J6) that provides a jumper-selectable 0–10 VDC or 0–20 mail input with a software programmable gain and offset adjustment (See Functions F-11 and F-12 in section 5).

- Note: If the position of the reference jumper is changed, the software does not recognize that the input signal has been changed from 0–10 VDC to 0–20 ma, or vice versa. Verify that calculations for lunctione F-11 (remote reference gain) and F-12 (remote reference offset) are correct before starting the drive.
- To change the remote speed reference jumper J6:

DANGER

AFTER DISCONNECTING INPUT POWER WAIT FIVE MINUTES AND CHECK. WITH A VOLTMETER TO INSURE THAT D-C BUS CAPACITORS ARE DIS-CHARGED. VOLTMETER SHOULD READ ZERO VOLTS D-C. FAILURE TO OB-SERVE THESE PRECAUTIONS COULD RESULT IN SEVERE BODILY INJURY OR LOSS OF LIFE.

- 1. Turn of power supplied to the control er.
- If not already cone, loosen the four (4) effecting ecrews and remove the cover from the controller.
- Verify at the + and terminals that the D-C Bus voltage is zero. (3) VDC. Refer to Figures 3.7 and 3.8.
- Locate jumper J8 on the Controller's Regulator Board. Refer to Figure S-4, 3-5 and 3-5.
- 5. Locate Pin 1 of jumper J8 on the Pegulator Roard.
- Move the jumper to the desired setting as shown in Figure 3-17.



Figure 3-17. J6 Jumper Settinge.

- After moving jumper (J6), ver 'y (with each move) the. Function F-11 (reference gain) and F-12 (reference offset) are correctly set.
- B. Make a written note of the jumper settings in the table labeled "Settings Table" on the back cover of this manual.

Optional Dynamic Braking Wiring

 Route the dynamic braking power and signal wring through the opening at the bottom of the controller as follows:

Refer to Figures 3-4, S-5, and 3-6.

- Size A Route dynamic power and signal leads through the bottom left opening of the controller base. Continue routing the signal leads up and eround the bracket assembly.
- Size 8 Roule dynamic power and eignal leade through the bottom center opening of the controller.
- Size C Route dynamic power and signal leads through the third opening (from the left) of the controller.
- Note: On size B and C controllers, continue routing the signal leads up, around the bracket szeembly, and through the support clip.
- Wire the Dynamic Braking (DB) Power leads to the controllers as follows. Refer to Figures 3-18 and 3-19.

For Model 1SU2ox controllers -

DB terminals 147 and 45 to controller power terminals +, - (DC Bus Volts).

For Model 1SU4xxx and 1SU5xxx controllers -

DB terminals 147 and 45 to controller power terminals $\pm_i = (DC BUS Volts)$

UB terminals 13 and 14 to controller power terminals + (10v), - (10v com)

 Wire the Dynamic Braking (DB) signal leads to the controllers as follows. Refer to Figures 3-18 and 3-19.

For all models: DB terminals 1 and 2 to controller control terminals 12 and 13.





3-27



Figure 3-19. Dynamic Braking Output Wiring For Models 1504xxx and 1SU5xxx Controllers.

3 29

4: Keypad and Display Operation

DANGER

ONLY QUALIFIED ELECTRICAL PERSONNEL FAMILIAR WITH THE CONSTRUCTION AND OPERATION OF THIS EQUIPMENT AND THE HAZ-ARDS INVOLVED SHOULD INSTALL, ADJUST, OPERATE AND/OR SERVICE IT. READ AND UNDERSTAND THIS MANUAL IN ITS ENTIRETY BEFORE PRO-CEEDING. FAILURE TO OBSERVE THIS PRECAUTION COULD RESULT IN SEVERE BODILY INJURY OR LOSS OF LIFE.

Keypad and Display Description

The controller contains a four-character display, eight indicator lights, and eix keys as shown in Figure 4-1. The display is used to inclicate power up self test, one of four display modes. program functions, program function values, and fault codes. The indicator lights (LFDs) give controller status on display modes, and mode of controller operation. Finally, the keys provide for the control and programming of the controller.

	$\Box \Box$		
	ΓЦЦ	Mace	Forward
П ЗРМ		(Futter)	Peverse
Vota	Program Forward	START	STOP RESET

Figure 4-1. SPS00 Controller Keypad and Display.

Indicator Lights

In general, the group of LEDs on the left indicate display modes while, the group on the right indicate operation modes. However, the fourth LED on the felt indicates whether the controller is under LOCAL or REMOTE control.

Display Mode LEDs

The four (4) LEDs on the left of the controller's keypad indicate the current display mode of oither RPM, %Load, Vors, or (when enabled) %. External speed reference. While the controller is running, the mode cisplayed and LED indication can be changed by using the MODE/ENTER key. (Refer to Modes of Controller Operation, later in this section for more information on changing display modes.) These LEDs are described further below.

 RPM LED The RPM/Engineering Unit LED indicates that the values being displayed, while the controller is unning, ere in units of HPM or some user unit")
 specified engineering unit to match en application. Used for display curcoses only, an engineering unit is a unit on a user defined scale based on the maximum output frequency of the crive. (See Section 5, Adjusting the Controller Functions, for more information.)

The 'RPM' mode is the drives default display mode and this LED will (luminate upon the init ellower up of the controller. However, once the display mode unit is changed (programmed), the controller will power up in the display mode unit active at the time it was turned off.

 %LOAD LHD: The '%LOAD' LHD indicates, while the drive is running, that the values being disclayed are in a percantage of full load ampa. When the drive is stopped, the value displayed will be zero.

 VOLTS LED: The 'VOLTS' LED indicates, while the drive is running, that the value being displayed is that of the drive surput voltage to the motor. When the crive is stopped, the value displayed will be zero.

- REMOTE IF the 'RPM", "%LOAD", "VOLTS" and "REMOTE' RETERENCE LEDs are it, the value displayed is the percentage (% external speed reference. This cisplay mode is speed only active when: reference)
 - The drive is in REMOTE mode (Function F-00 is enabled (ON)). (See Section 6, Adjusting line Controller Functions)
 - An external speed reference is connected to the Control terminal strip (See Section 3, Installation and Wiring)
 - Jumper J6 is properly selected (See Section S, installation and Wiring)
 - Function F 13 is enabled (ON): (See Section b, Adjusting the Controller Functions)
- Note: The last display mode chosen when the crive is privated down will then be the same one active when the drive is powered up again. Example: The display mode RPM is changed to %LOAD. The drive is powered down, and powered back up again. The display mode will remain as %LOAD.

Operation Mode LEDs

The group of four (4) LEDs on the right side of the controller's keyped, indicate that the controller is running (RUK LED), whether the Program Mode (PROC RAM LED) is active, or indicates the speed direction (either FORWARD or REVERSE). The fourth LED on the left indicates whether the control or REVERSE). The fourth LED on the left indicates whether the control or REVERSE. The fourth LED on the left indicates whether the control or REVERSE. The fourth LED on the left indicates whether the control or REVERSE. The fourth LED on the left indicates whether the control or REVERSE. The fourth LED on the left indicates whether the control or REVERSE.

DANGER

THE RUN LED MUST NOT BE USED AS AN INDICATION THAT THERE IS NO LINE VOLTAGE PRESENT IN THE CONTROLLER. VERIFY THERE IS NO VOLTAGE PRESENT AT THE D-C BUS TERMINALS (+) AND (-) BEFORE SERVICING THE CONTROLLER. FAILURE TO OBSERVE THIS PRECAUTION COULD RESULT IN SEVERE BODILY INJURY OR LOSS OF LIFE.

 PROGRAM LED: 	This LED is illuminated when the control or is in Program Mode. Program Mode is entered by pressing the MODE/ENTER key when the crive is stopped.
 FORWARD LED: 	This LED is illuminated, when the drive is running, to indicate that the requested rotation of the motor is in the Forward Direction.
REVERSE LED:	This LED is illuminated, when the controller is numning, to indicate that the requested rotation of the motor is in the Reversed Direction. However, when Function 17 is enabled: (=ON), the LED can not be Illuminated because the requested rotation of the motor is forced in the Forward direction. See Section 5, Adjusting the controller functions, for more information.
 REMOTE LED: 	The "REMOTE" LED incloses, when illuminated, that the drive is in remote operation and a following signals from the terminal strip. When the REMOTE LED is no. It, the orive is in LOCAL mode. Enabling Function F 00 (=ON) will illuminate the LED. (See Section 5. Adjusting the Controller Functions, for more information on function (0.)

FUN LED: This LED is Huminated when the drive is running (the controller is generating an output voltage and irequency).

Key Descriptions

Drive operation is controlled by six keys. These keys are described below.

- START key: The START key will start the controller when:
 - Power is applied to the controller
 - No faulte are active
 - Drive is in the LOCAL control mode.

When this key is pressed, the controller will accelerate to the last programmed frequency set point (speed). The frequency set point is the last displayed RPM (speed) or engineering unit value set from the up/cown arrow keys before the drive was turned off anc/or powered down. The crive will also display its last mode selection along with illuminating the appropriate display, rotation, and run indicating LEDs. The controller remembers and Lises all the functional settings and values that were enabled and selected prior to turning it off anc/or powering down.

Note: The controller will accelerate based on its programmed acceleration rate (Function F-01). (See Section 5, Adjusting the Controller Functions, for more information.)

If the acceleration rate is set to high, the motor will draw too much current resulting in a overcurrent (OC) IET Fault. When an IET Fault occurs, the drive will coast to a stop and the fault type code is flashed on the display. (See Section 7, Troubleshooting, Fault Codes, and Replacement Parts, for more information.

DANGER

AFTER DISCONNECTING INPUT POWER WAIT FIVE MINUTES AND CHECK WITH VOLTMETER TO INSURE D-C BUS CAPACITORS ARE DISCHARGED. FAILURE TO OBSERVE THESE PRECAUTIONS COULD RESULT IN SEVERE BODILY INJURY OR LOSS OF LIFE.

> STOP/RESET key: The STOP/RESET key provides three functions for the controller. If the controller is running, this key will turn off the controller. If the controller is stopped due to a fault, this key will clear the fault from the controller providing the cause of the fault has been removed from the controller. When the controller is in the program mode, this key will terminate the program operation.

> > THE STOP/RESET KEY WILL ALWAYS STOP THE CONTROLLER REGARDLESS OF THE SELECTED CONTROL MODE. (REMOTE OR LOCAL).

Note: The drive will perform a ramp to a rest stop (ramp down based on the set deceleration rate) or coast-to rest stop, depending on the value of Function F-16, when stopped under normal concisions.

If the deceleration is set too high, regenerative mather voltage may charge up the D-C bus voltage causing a high bus voltage (HU) IET Fault. When an IET Fault occurs, the controller will coast to a stop and the fault type code is flashed on the display. (See Section 7, Troubleshooting, Fault Codes, and Replacement Parts, for more information.)

MODE/

ENTER key:

The MODE/ENTER key provides three functions for the crive. If the drive is nurning, this key will change (or "toggle through") the active display mode (RPM, %_OAD, VOLTS, or REMOTE REFERENCE). If the controller is stopped, this key will select the drives program mode. Once in the crogram mode, this key will display and save a function value. The MODE/ENTER key functions in either REMOTE or LOCAL control modes. FORWARD/

REVERSE KEY:

The FORWARD/REVERSE key is used to select ("topgle between") the direction for motor rotation. The FORWARD or REVERSE LED will also illuminate to indicate the roquested rotation direction for the motor. However, when Function F-17 is enabled (ON), this key will not change the motor rotation because the controller is forced (programmed) only in the Forward direction. See Section 5, Adjusting the **Controller Functions**, for more information. This Key is only active in the LOCAL control mode.

- Note: The LED may illuminate before the motor turns in the new direction.
- UP/DOWN ARROW KEYS:

S: The UP and DOWN ARROW keys provide three functions for the controller. When the controller is running, whether in the LOCAL or REMOTE operation, they are used to increase or decrease the (internal) speed reference of the controller. When the controller is slopped and placed in the program operation mode, these keys will step through the controller functions (F-00 to ERR). The UP ARROW will increase the Controller Function number. After the selection of a controller function number. After the selection of a controller function function, these keys are used to controller function function in the selection of a controller function number. After the selection of a controller function function, these keys are used to change function status (DMO-E) or value (step incrementing).

Note: While in Remote operation mode, the changing of the crives internal speed reference, using the UP and DOW'N ARROW keys, will NOT have any operational effect on the controller. The controller's display will show the internal speed reference. In Hertz (Hz) while using the keys. A slight delay will be noticed before the controller's display returns to the display mode active prior to changing the internal speed reference.

> Holding the UP/DOWN ABROW keys for more than a few seconds will increase the scroll speed.

Power Up

When the power is applied to the drive, the display will show "SELF" and during this display period, the microcontroller will perform a self-tes, diagnostic routine. This is normal and the display should "change" to the RPM display mode, or whatever mode the controller was in when t was powered down last.

Modes of Controller Operation

Display Modes

There are three display modes (or four, if Remote Reference Display mode is active) available when the drive is powered up. These are: RPM, %i cad, and Volts. The current display mode (while the crive is running) is indicated by an illuminated LED. The factory default display mode is RPM.

To select a display mode (RPM, %LOAD, or VOLTS):

 While the drive is continue, preasitive MORE/ENTER key. The display will show WLEAD, and kine %LCAD. LED, will the it. Nachrime the MOS IP AND HE key is pressed, the best mode, le displayed and the approximate (LED, will light this value displayed and the approximate (LED, will light this value displayed and the approximate (LED, will light this value displayed and the approximate (LED, will light the value displayed and the approximate (LED, will light the value displayed and the approximate (LED, will light the value displayed and the approximate (LED, will light the value displayed and the approximate (LED, will light the value displayed and the approximate (LED, will light the value displayed and the approximate (LED, will light the value displayed and the approximate (LED, will light the value displayed and the approximate (LED, will light the value displayed and the approximate (LED, will light the value displayed and the approximate (LED, will light the value displayed and the approximate (LED, will light the value displayed (LED, will light the light the light the value displayed (LED, will light the light the light the value displayed (LED, will light the light the light the displayed (LED, will light the displ

A fourth display mode, "REMOTE REFERENCE DISPLAY", can be activated by Function F-13. The value displayed in this mode is the value of the external reference signal as 5–100% of the total scaled reference range. (Functions F-11 and F-12 are functions that "scale" the reference.)

If Function F 13 is enabled (or turned "ON") a fourth display mode choice can be activated by pressing the MODE/ENTER key once after the "VOLTS" display mode. All three LED's (RPM, %LOAD, and VOLTS) will be illuminated at the same time. Indicating the fourth display mode is active. Refer to Function F-13 for more information and examples for the REMOTE REFERENCE DISPLAY ENABLE Function.

To select the REMOTE REFERENCE DISPLAY mode: 1.0 7. HISSON MUTHING YO SHOUTS TROUTAN MOS THE CAOCHEM CER 400-DE INFORMATE AND MER CORDINE DOW THOSE X Press of up a tox fee up of C-16 be dealed to o down strtwtw(wire)-cretw(wire)-bab-babcle(+io-techonder)-right-cretwire)-cretw(wire)-cretw(wire)-babc-babcle(+io-techonder)-cretwire)-cretwi cretwire)-cret hisrocon in the second second A FIGS BOM SHALL BASK 5. Maye the up active key both the blab av shows "O6" T. Piess & BIOPHFOECKey In eth by PROBRAM Made. E Press the MORE/ENTER Rev unital three dealey made LEBEART the the cleb by block row by in the new ortent frences dischard:

Program Mode

The Program Mode allows function values to be viewed or adjusted. All functions (00-19), including the error log (ERB') are accessible in the Program Mode.

- Note: The PROGRAM mode can only be entered if the drive is stopped.
- To enter the Program Mode and access the functions:



To exit the Program mode:



Error Log

Localed after Function F-18, is an error og ("Err" is displayed) that stores the frat inree faulta that have occurred. The Program mode mus, first be entered to access the error og. Refer to Section 7, *Troubleshooting Fault Codes, and Replacement Parts* for more information on the Error Log and clearing the faults.

To access the Error Log:



5: Adjusting the Drive Functions

DANGER

ONLY OUALIFIED ELECTRICAL PERSONNEL FAMILIAR WITH THE CONSTRUCTION AND OPERATION OF THIS EQUIPMENT AND THE HAZ-ARDS INVOLVED SHOULD INSTALL, ADJUST, OPERATE, AND/OR SERVICE THIS EQUIPMENT. READ AND UNDERSTAND THIS MANUAL IN ITS EN-TIRETY BEFORE PROCEEDING. FAILURE TO OBSERVE THIS PRECAUTION COULD RESULT IN SEVERE BODILY INJURY OR LOSS OF LIFE.

> The drive offers 19 functions that can be either accessed or monitored, and 17 functions can be changed using the keypad. The first eighteen (18) functions are shipped from the factory with default settings. These settings must be adjusted for the application.

F=-00 Remote enable Ca., CFF N/A F=-01 Accel rate (see) 0.5-30.0 0.10 - 02 December (see) .6-30.0 0.10	
F01 Accel ette (see) 0.5-30.0 0.10 - 02 Dece mile (see) 10-30.0 0.10	OFF
- 02 Decemble (sec) .0-30.0 0.10	3.0
	0.0
F = 03 Winimum speed (1 z) 2.0 30.0 0.10	5.0
F-04 kaziman Speed (Hz) 30.0-240.0 0.10	82.0
F-05 Current limit (X) 10% 150% 1.0	150%
F-06 Torque Beast (8) 0% 10% 1.0	24
$\Gamma = 37.7/1.7$ (Dase speed) $30 = 240$ 1.0	H:I
1> FCB 3FV at base speed inwasa into	1750
F-09 IET output enclus (n-jm) N/4	201
T-10 Conier Lectency (NL2) 4,8,8 N/A	6
F-1 Serios tel quin (X) - 402-1002 - 0.10	300
F = 12 Remote ref office1 (ii) arg = 4 arg = a up	62
20 F-1 A Renda Ref. Bisplay Ends a - John N/A -	OT
F-14 des llema overtes 205-1005 1.0	1032
F-15 Det hernal 0/L enable - conjoin - K/A	CFF
F-16 Coost Stop English Ch.OFF K/A	Su
T-17 Reverse Disable Cn/01 N/A	CFF
F-18 Software version (X.84) K/A	Ficul with
5> ERR (Displayed code) ET fault codes	
FU High DC sta votoga o	and ikm
W ov 00 has sollege o	erif bee
CL Electronic transfer ev	budies.
GF Contro on High Tomp	oraturo
00 OvercurrentShort elitation (insun: faut
FL Externa function cas	detressed.

Figure 5-1. Factory Default Settings

DANGER

ONLY QUALIFIED ELECTRICAL PERSONNEL FAMILIAR WITH THE CONSTRUCTION AND OPERATION OF THIS EQUIPMENT AND THE HAZ-ARDS INVOLVED SHOULD INSTALL, ADJUST, OPERATE, AND/OR SERVICE THIS EQUIPMENT. READ AND UNDERSTAND THIS MANUAL IN ITS EN-TIRETY BEFORE PROCEEDING. FAILURE TO OBSERVE THIS PRECAUTION COULD RESULT IN SEVERE BODILY INJURY OR LOSS OF LIFE.

To change the value of a function:



F-00 - REMOTE ENABLE

Settings

OFF = Local Operation ON = Remote Operation

Description

This function selects the control mode for the drive. In LOCAL control the drive takes input from the keypad, while in REMOTE control the drive takes commands from signals on the terminal strip. If the controller is in REMOTE mode, the REMOTE LED will be lit.

Under LOCAL control the following keys can be used:

- START
- STOP/RESET
- FORWARD/REVERSE
- MODE/ENTER (when drive is stopped)
- Up and Down Arrow (when drive is in PROGRAM mode) In addition to these keys, the following signals are processed:
 - Remote IFT/Controller Hunning (see Figure 3-9)
 - Remote Function Loss (see Figure 3-9)
 - Remote Display (RPM, %LOAD, VOLTS)

Under REMOTE control the following keys can be used:

- MODE/ENTER (when drive slapped)
- Up and Down Arrow (when drive is in PHOGRAM mode).
- STOP/RESET

All remote signals (e.g., remote stop, start, etc.) wired to the terminal strip are processed. (See Figure 3.9.)

F-01 - ACCELERATION

Settings

Range of 0.5 - 30 seconda

Step Size

0.10 seconds

Description

The acceleration race is the amount of time to rame from stop to the programmed maximum speed setting (Function F-04).

If the selprint finducincy (last induced local speed from the up/down arrow key) is less than the maximum speed setting, the time to ramp to that astpoint will be proportionally less than the actual acceleration rate setting. If maximum speed equals 60 Hz and accel time equals 4 seconds, it will take 2 seconds to ramp to a frequency setpoint of 30 Hz (and 1 second to ramp to 15 Hz).

If the acceleration rate is set too fast, the controller may trip out on an overcurrent condition. (See Section 7 if this condition exists.)

F-02 - DECELERATION

Settings

Range of 1.0 to 30 seconds

Step Size

0.10 asconde

Description

The deceleration rate is the amount of time to ramp from the programmed maximum speed setting (Function F-04) to a stop.

The deceleration time is also proportional to the last set frequency. (See Function F-01, Acceleration.)

If the deceleration rate is set too tast, the drive may bip out on a high bus tault. (See Section 7-11hs condition exists.)

DANGER

THE SP500 CONTROLLER IS INTENDED TO OPERATE THE MOTOR AT A PREDETERMINED MINIMUM SPEED UNLESS DISCONNECTED FROM THE POWER SOURCE. TO PREVENT INADVERTENT CONTACT WITH OPERAT-ING EQUIPMENT, THE USER MUST VERIFY THAT THE MOTOR OUTPUT SHAFT WILL ROTATE AT ALL COMBINATIONS OF LOAD AND OUTPUT SPEED REQUIRED BY THE APPLICATION. FAILURE TO OBSERVE THIS PRE-CAUTION MAY RESULT IN SEVERE BODILY INJURY OR LOSS OF LIFE

Settings

Range of 3 - 30 Hz

Slep Size

0.10 Hz if maximum speed is lees than 100 Hz 0.25 Hz if maximum speed is greater than or equal to 100 Hz.

Description

Minimum speed is the user reducated minimum output frequency value of the controller. This speed (Hz) reference must always be lower than the maximum (Hz) setting (Function F 04) of the controller. The base or running speed setting (base speed, Function F-07) will always be a value between this minimum and the maximum speed (Hz) reference.

WARNING

THE USER IS RESPONSIBLE FOR ENSURING THAT DRIVEN MACHINERY, ALL DRIVE-TRAIN MECHANISMS, AND PROCESS LINE MATERIAL ARE CA-PABLE OF SAFE OPERATION AT AN APPLIED FREQUENCY OF 100% OF THE MAXIMUM FREQUENCY (UP TO THE FREQUENCY OF F-04). FAILURE TO OBSERVE THIS PRECAUTION COULD RESULT IN BODILY INJURY.

Settings

Range of 30 - 240 Hz

Step Size

0.10 Hz 1 max speed is less litan 120 Hz 0.25 Hz 1 max speed is greater than or equal to 100 Hz

Description

The maximum speed is the user requested maximum output frequency value of the controller. This maximum speed reference (Hz) must always be higher than the minimum (Hz) sotting (Function F-02) of the controller. The base or running speed setting (case speed Function F-07) will always be a value between the minimum reference and this maximum.

F-05 - CURRENT LIMIT

Settings

Range of 10% to 150% of rated controller current

Step Size

1.0%

Description

This function provides the means to limit motor output torque during run or acceleration. When output current attempts to exceed the preset current limit, motor speec is maintained or reduced, or deceleration time is extended. If current limit is set foo low, an "OC" (Dveroument) fault may occur.

F-06 - MANUAL TORQUE BOOST

Settings

Range of 0 - 10%

1 = 1% 2 = 2% 10 = 10%

Step Size

1.0%

Description

Torque boost is required to offset the voltage drop of the A-C motor at low speeds. For friction roads, or high inertia loads, a high starting torque level may be needed. Manual torque boost is only effective at speeds, ower than one-half of base frequency. See Figure 5-2.

When adjusting this function, start with the celauit setting of 2%, and gradually increase the adjustment until satisfactory motor operation is reached.



Figure 5-2. Manual Torque Boost Adjustable Range.

F-07 - V/HZ (Base Speed)

Settings

Range of 30 - 2/0 Hz

Step Size

1.0 Hz

Description

The volts/hertz (sature allows the drive to maintain a constant volts/hz ratio, thus providing constant torque at any frequency. The frequency value entered to establish the volts/hertz curve is the base speed at maximum output voltage. See Figure 5-3.



Figure 5-3. Volta/Hertz Curve.
F-08 - RPM AT BASE SPEED

Settings

Range of 1 - 9939

Step Size

1.0 unit

Description

This feature provides you, with the ability to scale the display to any desired engineering unit to match your adecitic application. Although Function F-08 is called "RPM at Base Speed", any engineering unit can be used. You could set it up as Hertz, Feet/Min, or any other unit needed. The value scaled in this function is what is displayed when looking at the Display Mode, "RPM".

The value entered into Function F-D8 is the maximum value (RPM, or any other engineering unit) to be displayed in the RPM display mode, when the controllar is running at base speed (Punction F-07).

Exemple 1: You want 1750 BPM to be the displayed value when the controller is running at a base speed of 60 Hz. You enter 1750 into Function F-08.

The actual displayed value for the RPM display mode, is a acaled value of what is entered for Function F-08:

Example: You want 1750 RPM to be the displayed value when the controller is running at a base speed of 60 Hz. You are presently running at 30 Hz. The RPM display mode value is scaled as follows:

Present Operating	×	F-08 (You entered 1750)
Speed (Hz)		F-07 (Base epace of 60 Hz)
SOHZ	×	<u>1750</u> = 875

In this example, when you are running at 30Hz, the RPM display mode will show 875.

Example 2: You want 800 feat/minute to be the displayed value when the controller is running at a base speed of 60 Hz. You enter 850 into Function F-08.

The actual displayed value for the RPM display mode, is a acaled value of what is entered for Function F-08:

Example: You want 800 feel/inin, to be the displayed value when the controller is running at a base speed of 60 Hz. You are presently running at 40 Hz. The RPM display mode value is scaled as follows:

Present Operating Speed (Hz)	x	F-08 (You entered 800) F-07 (Base speed of 60 Hz)
40Hz	x	<u>800</u> - 533 60

In this example, when you are running at 40Hz, the RPM display mode will show b33 feet/minute.

F-09 - IET (INSTANTANEOUS ELECTRONIC TRIP) OUTPUT ENABLE

Settings

ON - IET fault incleation

OFF = Controller is running indication

Description

This function selects whether the remote output signal (at terminals 14-16) represents the state of an active fault or that the controller is currently running. If the signal represents an active fault (F 09 – ON), then it will remain asserted until the fault is cleared by pressing the "STOP/RESET" key (in LOCAL operation) or by using the remote IET reset signal (for REMOTE operation). Refer to Figure 3-8.

If the signal represents controller running (F-09+OFF), then the signal is only asserted when the RUN LED is till.

F-10 - CARRIER FREQUENCY

Settings

Range of 4 - 8 KHz

4 = 4 KHz = 100% output amps 6 = 6 KHz = 90% output amps 6 = 8 KHz = 60% output amps (default setting)

Description

The carrier frequency can compensate for accustic noise, heating, and other current problems by adjusting the switching frequency of the transistors in the inverter section.

The carrier frequency controls the width of the pulse and keeps the current smooth to the motor.

Keeping the carrier frequency at 4kHz will maximize the continuous power rating of the crive and generally still have an acceptable accustic noise from the motor. Increasing the carrier frequency will alleviate the accustic noise, but in some applications can result in derating of the controller output amps. (Refer to Tables 2-1 through 2-5 for the cerated ratings at the various carrier frequencies of the controllers.)

F-11 - REMOTE REFERENCE GAIN

Settings

Range of 60% - 100% of full scale maximum reference

Step Size

0.10%

Description

The remote reference gain scaling is used to scale the **maximum** remote speed reference to match external equipment. Normally, the maximum speed reference (amount of reference at maximum speed, F-04) is either 10 VDC or 20 ma. The reference gain is used to scale the speed reference to another value (for example 9.5 VDC or 19 ma). Enter this function in percent (%) of full scale reference. To calculate the scaled reference, use the following equations:

If using a 0 - 20 ma remote reference:



Example 1. If the remote speed reference is 0–20 maland the maximum reference required is 19.2 ma, scale as follows:

19.2	ĸ	100	-	96% gain
20				0000657493

If using a 0 - 10 VCC remote reference;

1 Jacob A. Jacob Anton B	Addread and the second and the secon	
1. 6. Addie A. 6. 281 1. Aufr. 1.	2 Provident 123 / 101 = 13 / 29(1)	CC-Distence)
". '. '. '. SAIA & Ada'Es	MARINAL CONTRACTOR CONTRACTOR	300030000030
0.0000000000000000000000000000000000000		

Example 2. If the remote speed reference is 0 – 10 VDC and the maximum reference required is 9.5 VDC, scale as follows:

> <u>9.5</u> x 100 = 95% gain ⁻0

F-12 - REMOTE REFERENCE OFFSET

Settings

Pange of 0% - 40% of full scale minimum reference

Step Size

0.10%

Description

The remote reference offset scaling is used to scale the minimum remote acceed reference to match external equipment. Normally, the minimum speed reference (amount of reference at minimum apaed, F-03) is either 0 VDC or 0 mA. The reference gain of set is used to scale the minimum zero speed reference to another value (for example 0.4 VDC or 4 mA). Enter this function in percent (%) of full scale reference to be offset from minimum speed. To calculate the scaled minimum reference, use the following equations:

If using a 0-20 ma remote reference;

. Destrute Ann Scient Office (ma). X 100 ∓ % bitset

Exemple 1. If the remote speec reference is 0-20ma and the offset from minimum speed required is 4 ms, acale as follows: <u>4</u> x 100 - 20% offset 20

If using a 0-10 VDC remute reference;

Besided Zerb Speed Olber, NOCI	X 100' A S/officat
fieleperise Ronge (19)	

Example 2. If the remote epeec relevance is 0–10 VDC and the oilset from minimum speed required is 0.4 VDC, scale as follows:

> 0.4 x 100 - 4% offset 10

F-13 - REMOTE REFERENCE DISPLAY ENABLE (4TH DISPLAY MODE)

Settings

Displays the external speed reference as 0 - 100% of the ecaled reference range.

ON = 4.h display mode is enabled and will display the external speed reference when the REMOTE display mode is chosen. (Function F-00 = ON)

OFF – 4th display mode is disabled. The external speed reference will not be displayed.

Description

This function enables a 4th display mode which allows the display of the current value of the external speed reference as a value in percent (%).

When enabled (Function F-13 – ON), the 4th display mode is activated by pressing the MODE/ENTER key (while the drive is running) and all three display LEDs are illuminated at the same time.

When the REMOTE REFERENCE Display Mode is chosen, the display will show the external reference as 0 – 100% of the speed reference range as specified by Function F-11 (remote gain) and Function F-12 (remote offset). Refer to Section 4. on ***To select the REMOTE REFERENCE Display Mode.***

Exampla:

10 \pm M vericle gain from the must reference = 65% stric. \pm M vericle altertification of the transference = 6%, the depthy would include "NUD" it the external reference = \pm 9% M of an evolution cape 0.0 M to be external reference \pm 9.5 VDC.

F-14 - ELECTRONIC THERMAL OVERLOAD

Settings

20% - 100% rated correct.

Step Size

1%

Description

This function should be adjusted if the motor amperage rating is less than the controller emberage rating. Using the formula below, calculate the setting level as a percentage of maximum continuous current;

ł	ļ	Ś	ł	į	į	ŝ	ŝ		2	ŝ	i	÷	÷	Ń	ŝ	à	Ĥ	Ě	Å	Ĺ	ŝ	ź	Ř	įψ	'n	e	ıţ	ł	Ş	ì	Ŷ	ł		ġ	3	ŝ		ŝ	í	
6	ĝ	ŝ	į	ŀ	ŝ	ŝ	1	ŝ	ĝ	Ş	1	Ş	ņ	1	Ņ	9	ŀ	P	ų	Þ.	H,	Ę	Ņ	4	ņ	\$	ţ	4	4	ŝ	ł	ġ	1	1			ŝ	ŝ		

The motor full load current can be taken from the motor nameptate, and the controller rated output current can be taken from Table 2-1 through 2-5 as applicable (or the controllar nameptate).

CAUTION: Function F-15 should be set equal to "ON" (ENABLED). Failure to be serve this precaution could result in damage to, or destruction of the motor and controller.

F-15 - ELECTRONIC THERMAL OVERLOAD ENABLE

Settings

OFF - No electronic thermal overload protection (factory default).

ON - Electronic thermal overload protection is active.

The electronic thermal overload enable is used to simulate the functionality of a motor thermal switch, protecting the motor from overheating due to excessively high current in a short period of time.

If this function is ON, the controller will trip if the thermal overload time is exceeded (80 seconds at 160% of F-14). When an external thermal switch (an alarm, or warning light, etc.) is wired into the Function Loss circuit (terminals 10 and 11), this function should be set equal to "ON". Heter to Figure 3-9, Control Terminal Strip Wirting.

CAUTION: This function should be set equal to "ON" (ENABLED). Failure to observe this precaution could result in damage to, or cestruction of the motor and controller.

WARNING

THE CONTROLLER IS NOT EQUIPPED WITH A COAST-STOP PUSHBUTTON. THE USER MUST INSTALL A HARDWIRED, OPERATOR-ACCESSIBLE PUSH-BUTTON THAT PROVIDES A POSITIVE INTERRUPT AND SHUTS DOWN THE DRIVE. (USE TERMINALS 10 AND 11. SEE FIGURES 3-4, 3-5, 3-6 AND 3-16.) FAILURE TO OBSERVE THIS PRECAUTION COULD RESULT IN BODILY INJURY.

Settings

ON = Enable Coast-Stop (Coast-to-rest)

OFF - Disable Coast Stop (Ramp-to-reat)

Description

This function, when enabled, will permit a coast-to-rest stop, instead of the defaulted ramp-to-rest stop.

F-17 - REVERSE DISABLE

Settings

ON - Disable Reverse (from LOCAL and REMOTE control)

OFF - Enable Reverse (This is the default setting)

Description

When this function is ON, the drive will not be able to turn in the reverse direction. Also, when the crive is in LOCAL mode, pressing the Forward or Reverse keys, will have no effect on drive operation. The Forward LED will always display ON, and the Reverse LED will always display OF? When the crive is in REMOTE mode, any wiring to terminal 9 of the Control terminal strip (see Figure 3-9) is ignored, and the drive will only run in the forward cirection.

Hegerdiess of what is entered or changed, when this function is ON (reverse is disabled), the drive will be forced in the forward direction.

F-18 - VERSION INFORMATION

Settings

XXX = Version Number

Description

When the diaplay shows "F-18", press the MODE/ENTER key to show the current revision. This function cannot be changed.

ERR - ERROR LOG

Peter to Section 7, "Troubleshooting, Fault Codes, and Replacement Parts" for information on the Error Log that follows Function F-18.

ONLY QUALIFIED ELECTRICAL PERSONNEL FAMILIAR WITH THE CONSTRUCTION AND OPERATION OF THIS EQUIPMENT AND THE HAZ-ARDS INVOLVED SHOULD INSTALL, ADJUST, OPERATE, AND/OR SERVICE THIS EQUIPMENT. READ AND UNDERSTAND THIS MANUAL IN ITS EN-TIRETY BEFORE PROCEEDING. FAILURE TO OBSERVE THIS PRECAUTION COULD RESULT IN SEVERE BODILY INJURY OR LOSS OF LIFE.

Check the Installation

DANGER

AFTER DISCONNECTING INPUT POWER WAIT FIVE MINUTES AND CHECK WITH A VOLTMETER TO INSURE THAT D-C BUS CAPACITORS ARE DIS-CHARGED. THE VOLTMETER SHOULD READ ZERO UDC. FAILURE TO OB-SERVE THESE PRECAUTIONS COULD RESULT IN SEVERE BODILY INJURY OR LOSS OF LIFE.

- If an input disconnect is installed, make sure it is in the OFF position.
- Verify the D-C bus volts is zero. See Figures 8-7 and 3-8. Even if the bus LED is OFF, always verify that the D-C bus is discharged.
- Make sure the controller interlocks installed around the oriven machine are operational.

WARNING

THE SP500 CONTROLLER IS NOT EQUIPPED WITH A COAST-STOP PUSH-BUTTON. THE USER MUST INSTALL A HARDWIRED, OPERATOR-ACCESSI-BLE PUSHBUTTON THAT PROVIDES A POSITIVE INTERRUPT AND SHUTS DOWN THE DRIVE. (USE TERMINALS 10 AND 11, SEE FIGURES 3-4, 3-5, 3-6 AND 3-18.) FAILURE TO OBSERVE THIS PRECAUTION COULD RESULT IN BODILY INJURY.

4. Verify that the user-installed atop pushculton is operational. When pushed, it should shut down the controller. The factory-installed jumper should not be connected to terminals 10 and 11 for the stop pushbutton to work. Remove this jumper.

CAUTION: Make sure electrical commons are not intermixed in the controller. Failure to observe this precaution could result in carriage to, or destruction of the ecuipment.

- 5. Remove any debris from around the controller.
- 5. Check that there is adequate clearance around the controller.
- Check and verify theil the wiring to the control terminal strip and power terminals is correct (Figures 3-4, S-5 and 3-6).
- Check that the terminals are tightened property to the appropriate torque specifications given in Tables 3-8, 3-7 and 3-9.
- Check that user-supplied branch circuit protection is installed and correctly raced.
- 10. Check that the incoming A-C power is rated correctly.
- 11. Check the motor installation and length of motor leads.
- Disconnect any power correction capacitors connected to the motor.
- Uncouple the motor from any driver machinery to initially atart the controller;
- Check that any motor thermal switch and the controller's electronic thermal overload (Function F 15) is set (enabled).
- Check that the rating of the transformer (if used) matches the controller requirements, and is connected for the proper voltage.
- 16. Verify that a property sized ground wire is installed and that a suitable earth ground is used. Check for and eliminate any grounds between the motor frame and the motor power leads. Verify that all ground leads are run unbroken.

THE SUBSEQUENT STEPS REQUIRE ROTATING PARTS AND/OR ELEC-TRICAL CIRCUITS TO BE EXPOSED. STAY CLEAR IF UNIT MUST BE RUN-NING OR DISCONNECT AND LOCKOUT OR TAG POWER SOURCE IF CON-TACT MUST BE MADE. FAILURE TO OBSERVE THESE PRECAUTIONS COULD RESULT IN SEVERE BODILY INJURY OR LOSS OF LIFE.

DANGER

AFTER DISCONNECTING INPUT POWER WAIT FIVE MINUTES AND CHECK WITH A VOLTMETER TO INSURE THAT D-C BUS CAPACITORS ARE DIS-CHARGED. FAILURE TO OBSERVE THESE PRECATUIONS COULD RESULT IN SEVERE BODILY INJURY OR LOSS OF LIFE.

- 1. Jurn OFF, lock out and tag power to the controller.
- Connect a voltmeter to terminals (+) and (-), and verify that the controller bus voltage is 0 VOC. See Figures 3-7 and 3-3.
- 3. Uncouple the driven equipment from the motor; if possible.
- Turn the power ON. With a voltmeter at terminals (+) and (-), observe a voltmeter reacing for an approximate no load D-C bus voltage value on the control or as listed in Table 6.1. (D.C bus voltage readings are effected by the incoming line characteristics such as, amplitude, frequency, harmonic distortion, etc.)

A-C Input Line Voltage	D-C Bus Voltage (No Load) ⁽¹⁾				
Model 18	U2xx Controlers				
230 VAC	323 V				
Model 19	2.14xxx Controllers				
360 VAC	635 VOC				
400 VAC	563 V JC				
415 VAC	585 VDC				
440 VAC	620 V 3C				
460 VAC	646 VOC				
Model 19	USion Controllers				
575 VAC	81* V3C				

Table 6-1. D-C Bus Voltage Value.

(1) D-C No Load Voltage readings are approximate values and will vary greatly depending on the characteristics of the incoming A-C voltage line.

- Check all function settings (Section 5) and verify that they are set correctly.
- Press the START key. The controller should ramp to the preset maximum speed (F-04). The mater should ramp up at the acceleration rate (F-01) until it reaches the preset Hz.

AFTER DISCONNECTING INPUT POWER, WAIT FIVE MINUTES AND CHECK WITH A VOLTMETER TO INSURE THAT D-C BUS CAPACITORS ARE DIS-CHARGED. THE VOLTMETER SHOULD READ ZERO UDC. FAILURE TO OB-SERVE THESE PRECATUIONS COULD RESULT IN SEVERE BODILY INJURY OR LOSS OF LIFE.

> While the controller is in the BUN mode, check the display modes and verify that VOLIS and %LOAU are reading correctly.

For Remote: If using a remote speed reference check that the speed reference (F-13 must be "ON") is correct (0–10VDC, or 0–20mA). All three display mode LEDe should be fill to indicate that the remote reference is currently being displayed. The fourth display LED (REMOTE LED) should also be it.

Take into account any values set into F-11 (Bernote Beference Gein) and F-12 (Bernote Reference Offset) that have scaled the speed reference.

- RPM can size be monitored by using the RPM display mode.
- Verify the direction of the motor shaft rotation. Press the STOP key to stop the controller.
- 10. If the direction of shaft rotation is incorrect, change as follows:
 - a. Wait until motor has completely stopped.
 - Turn OFF, lock ou, and tag power to the controller.
 - Verify that the D C Bus voltage (at terminals + and -) is 0 VDC. See Figures 3.7 and 3.8.
 - Briverse any of two of the three meter power leads (U.V. or W).
- 11. Jum the Power ON.
 - 12. Press the START key.
- 13. Using the up or down arrow keys, change the maximum speed setting (F-04) and run the motor without any load across the speed range. If the motor is unloaded and does not operate satisfactory, check the function settings in Section 5. If it does operate satisfactory, go to Step 14.
- Tum QEF, lock out and tag power to the controller. Verify that the D-C bus has discharged to 0 VDC at terminals (+) and (+). See Figuree S-7 and 3-8.
- 15. Couple the driven equipment to the motor,
- 16. Turn Power ON.
- 17. Press the controller START key.
- Run the controller across the required speed range under load. If the indior does not rotate at minimum speed, increase manual torque boost (F.06).

- 19. If the controller operates the motor property:
 - Turn OFF, lock out and tag power to the controller. Verify that the D-C pus has discharged to 0 VDC at terminate (+) and (+).
 - b. Replace the controller cover and secure.
 - Make a note of final function softings in the "Settings Table" in the back of this manual.

If the controller does not operate the motor property:

- Refer to Section 7, Troubleshooting, Fault Codes and Replacement Parts 1 any fault indications occurred during startup.
- b. Verify function settings again.

Cover Installation, NEMA 4X (Indoor Only)/12

In order to maintain integrity of the NEMA 4X/12 enclosure, some care must be taken when re-installing the cover:

- Run the (4) captive screws down, sequentially, to ensure even compression of the gaskets. <u>Do not</u> exceed 20 inch-pounds of torque on these screws.
- When the cover has been properly tightened, there should be very little gap between the cover and keypad.

7: Troubleshooting, Fault Codes, and Replacement Parts

DANGER

ONLY QUALIFIED ELECTRICAL PERSONNEL FAMILIAR WITH THE CONSTRUCTION AND OPERATION OF THIS EQUIPMENT AND THE HAZ-ARDS INVOLVED SHOULD INSTALL, ADJUST, OPERATE, AND/OR SERVICE THIS EQUIPMENT. READ AND UNDERSTAND THIS MANUAL IN ITS EN-TIRETY BEFORE PROCEEDING. FAILURE TO OBSERVE THIS PRECAUTION COULD RESULT IN SEVERE BODILY INJURY OR LOSS OF LIFE.

System Operation

Refer to Figuree 7-1 and 7-2, System Block Disgrams.

The SP500 uses the conventional Inverter bridge to transfer energy from the A-C input line to the A-C motor. The A-C line voltage is rec.if ed through the input diode module which in turn generates the constant D-C cus voltage.

A large bus capacitor across the D-C bus smooths the D-C bus voltage and bulfers the current flow to the motor. The six IGBTs and the associated dindes convert the constant D-C voltage into PWM (pulse width modulated) waveforms.

The SP500 uses the Volts/Hz constant control scheme which generates constant primary flux over the variable speec range of the motor so that forque is linearly produced in proportion to the slip of the motor.









Fault Codes

The SP500 controller discleys fault (ERR) codes which assist in troubleshooting if a fault should occur. If a fault condition occurs while the drive is running, the drive will coast-to-rest and the fault type is flashed on the display as a 2-digit alphabetical code.

The lauft is then entered into the error log. The error log is accessible after Function F 18, and is designated as "CPR" on the disclay.

How to Access and Read the Error Log

Fault codes are entered into the error log in sequential order if more than 1 fault should occur. The first error to occur will be flashed on the disclay, and two more errors will then be logged into the error log. (The error og must be eccessed to see them). After three faults have occurred, no more subsequent faults will be entered into the error log.

The faults entered into the error log are numbered sequentially. If first an overcurrent fault were to occur, it would be in the error log as "1=OC". If next a thermal overload fault were to occur (and the first error was not yet cleared). It would be in the error log as "2=OC", and so on.

The last fault to occur will appear first when accessing the error log. For example, if the last fault was a low bus fault, and the error log has 3 entries, then the error log would display "3+LU" when the error log is first accessed.

Each 'ault in the error log can be cleared by pressing the STOP/RESET button until "Err" is displayed again.

Faults are retentive to the error log if a power loss occurs.

Table 7-1 lists fault codes, descriptions, causes, and actions if a fault occurs.

To access the Error Log:



AFTER DISCONNECTING INPUT POWER WAIT FIVE MINUTES AND CHECK WITH A VOLTMETER TO INSURE THAT D-C BUS CAPACITORS ARE DIS-CHARGED. THE VOLTMETER SHOULD READ ZERO UDC. FAILURE TO OB-SERVE THESE PRECAUTIONS COULD RESULT IN SEVERE BODILY INJURY OR LOSS OF LIFE.

Before servicing the controller:

- 1. Turn OFF, lock out and tag power to the controller.
- Verify that there is no voltage entering the controller at terminals R(L1), S(L2), and T(L3).
- Verify al D-C Bus terminate (-) and (-) that the D-C Bus has failen to zero VBC. This will take a few minutes to drop to a safe level. (Even 7 the bus LED is OFF, always verify that the D-C bus is discharged.) See Figure 7-8 and 7-4 for location of the D-C Bus.



Figure 7-3. D-C Bus Terminals on Model 1SU2xxx Controllers.



Figure 7-4. D-C Bus Terminals on Model 1SU4xxx and 1SU5xxx Controllers.

Table 7-1. Fault Code Action List.

HU:	High bus voltage condition.								
Cau	50	Ac	tion						
The sbox threa	D-C bue charged re the electronic trip enold evel.	The error will not clear until the bus fails below the high bus level.							
1.]	Jecel rate too tast Function 2).	1a 1b	I ower the decel rate (Function F 02). Refer to Section 5. . Install the optional DB kit.						
2. 1	Starting the crive into a forward running load that has a high inertia.	2.	install the optional DG Kit.						
LUI: 1	Low bus voltage condition.								
Cau	BE:	Ac	lion						
The tha e level	D-C bus has fallen below ectronic trip low threshold	早ら行任他を	es the STOP/RESET tton (or a REMOTE IET reset remote operation). a error will not clear until i input line voltage is within proper range. This may e a few seconds.						
1. 1	Lase of input power.	1.	Check incoming power.						
2. 1	Low line voltaga.	2.	Check incoming power.						
Notr	If a line dip or momentary p level is able to fee back to the drive will automatically running when the fault occ perform above steps, or pr	the reat urre	er loss occurs, and the bus proper range within 500 ms, art (If the drive was already c). If the drive does not restart, the STOP/RESET key.						
OL:	Electronic thermal overload								
Çau	90	Ac	tion						
The over exce profi incto exce 9.80	electronic therma load htp level has been eded. This tault sets the erive and ar from overheating due to selve current within edified certod.	Pri bu for Th int rate far	ess the STOP/RESET ton (or a REMOTE IET reset remote operation). a error will not clear till the drive has agrated to the proper ige. This may take a viseconds.						
1. 	The Current Limit Setting (Function F-05) is not set correctly.	1.	If current limit level is too low relative to load, increase the current limit level.						
2. 1	Function F-14. Electronic Thermal Overload is not set correctly to match the motor and controller complication	2.	Verify the value of Function F-14. Refer to Section 5.						

Table 7-1. Fault Code Action List (Continued)

Oł	: Thermostal/Drive Overload		
Ça	use	Ac	tion
Th ha the con in	e internal thermoalat s caused a trip ti indicates cessive temperatures the controller.	Pre (or Th intr ran	es the STOP/RESET button a IET RESET for remote operation). He error will not clear until the email drive temperature is back within tige. This may take a few seconds.
1,	The drive derating specifications are sx- caeded. See Table 2-1.	1.	Re-check the application and change the carrier frequency of Function F-10. Refer to Section 5.
2.	The ambient temperature of the controller is exceeded.	2.	Check the location site and move the controller to a cooler area.
o	C: Overcurrent		
Ce	1088	Ac	tion
Th co dri exo	e current rating of the ntroller (>200% rated ve current) has been ceeded.	Pre (or ren	ess the STOP/RESET button IET RESET for note operation).
Th	is fault can be caused by any of	the f	alawing conditions:
•	Short in drive outputs	1.	Verify that the input and autput wiring to the drive ans property connected.
2.	Ground fault condition	2:a	Verify that the input and output winn to the crive are properly connected. See Figures 3.4 and 3.5 pr 3.6.
		5P	Verify that the output winng to the motor is not connected to Ground or any other voltage aburca. See Figures 3-4, 3-6, or 3-6.
5.	netantarieque overcument resulting in greater than 200% rated drive current.	3a. 3b.	Increase acceleration or deceleration true (F-01 or F-02). Increase current limit level, if too low relative to the load. (Change in 5% increments.)
FL	: Function Lose		
Ca	use	Ac	tion
1h 68 1	e remote (external) func- n loss signal has been serted. (Terminals 10 and .)	Prebui tor Thi the un	ass the STOP/RESET licen (or IET RESET remote operation), is fault will not clear until i function loss signal is asserted.
đê	The externel equipment connected to the function loss terminals has failed, or is giving repeated stop requests.	18. 16.	Check the external equipment wired to the remote function loss terminals (10 and 11). Check function loss connections.

		Quantity per Horsepower						
Description	Part Number	3	5					
Fan Assembly	616161-R	1	1					
	615°61-S	1	1					
NEMA 4X Cover/Gasket	806522-1R	1	1					
NFMA 1 Cover	805508-2B	1	1					
Membrane Switch (Keypad)/Bracket	709511-18	1	1					
Regulator PCB	0-56837-210	1	1					
Capacitor PCH	0-56931-050	1	1					
Fan Assembly Internal	616159-1R	1	1					

Table 7-2. Replacement Paris List For Model 15U2xx Controllers (1)

Note: 1. Replacement parts is not recommended, and parts are not available for 1 and 2 Hp Model 1SU2xor controllers.

Table 7-3. Replacement Parts Lis	For Model 1504bccc Controllers
----------------------------------	--------------------------------

		Quantity per Horsepower					
Description	Part Number	1	2	3	6	7.5	10
Fan Assembly	616161-R	-		1	1	1	1
	615161-\$	-	-	-	-	٦.	1
NHMA 4X Cover/	306512-1H	1	1	1	1		-
Gasket	806522-1H	-	-	-	-	- ۲	1
NEMA 1 Cover	306504-2H	1	1	1	1	-	-
	805508-2H	-	-	-	-	1	1
Membrane Switch	709507-1H	1	1	1	1		-
(Keypad)/Bracket	709511-1H	-	-	-	-	- ۲	1
Regulator PCR	0-56917-210	1	1	- t	1	1	1
Capacitor PCB	0-56914-030	-	-	1	-		-
	0-56914-050	-	-	-	1	-	-
	0-66919-070	-	-	-	-	-1	-
	0-56919-100	-		-	-		1
Fan Assembly Internal	616159-1R	1	1	1	1	1	1

		Quantity per Horsepower					
Description	Part Number	1	2	3	5	7.5	10
Fan Assembly	61616'-R	-	1	- 1	1	1	1
	615161-8	-	_	-	-	1	1
NEMA 4X Cover/	805612-1R	1	1	1	1	_	-
Gasket	805522-1H	-	-	-	-	1	1
NEMA 1 Cover	805504-2R	1	1	1	1		-
	805506-213	-	_	-	-	1	1
Membrane Switch	709 5 07-1 R	1	1	1	1	-	-
(Keypad)/Bracket	709511-1B	-	-	-	-	1	1
Regulator PCB	0-56927-210	1	٦.	1	1	1	1
Capacitor PCB	0-56972-055	-	-	1	1	-	-
	0-56938-055	-	-	-	-	1	1
Fan Assembly Internal	615159-1R	-	-	-	1	1	1

Table 7-4. Replacement Parts List For Model 1SU5xxx Controllers

8: Glossary of Common Terms

Accelerating Torcue	An increase in torque (force) generated by a motor in order to achieve running speed.
Adjusting the Unive Functione	The process by which the user can select and adjust one or all of the 17 programmable functions. (Refer to Section 5.)
Altitude	The height above sea level were the drive is installed.
Base Speed	Nominal motor speed rating in RPM on the motor's nameplate. Found by subtracting the alip speed from the synchronous speed at base frequency.
Base Frequency	Motor nameplate frequecy rating
Carrier Frequency	The switching frequency at which an IGBT inverter bridge converts a D-C voltage into a PWM voltage.
Caution	Alerts a person that, if procedures are not followed, damage to or destruction of, equipment could result.
Constant Torcue	A torque (force) characteristic independent of motor speed.
Controller	The term substituted throughout this manual for the "SP500 A-C V+S Controllers".
CSA	The abbreviation for the Canadian Standards Association.
Current Feedback	A current signal used by the microcontroller to control the ocerating current of the drive.
Danger	Alerts a person that high voltage is present which could result in severe bodily injury or loss of life.
Decelerating Torque	The torque (force) generated, by the decrease in motor and load kinetic energy, required by the motor and load to reach its final (slower) synchronous speed condition.
Drve	The reference to the controller and the motor combined as one system. The reference has also been used for another name for the controller.
Dynamic Braking	A braking technique in which the kinetic energy is converted into electrical energy and dissipated as heat energy via a resistor or other means.

Frequency Setpoint	The frequency value stored in memory (either by the local or remote means) within a given frequency range of the drives output voltage. Used to set the speed of the motor.
Hz	The Abbreviation for Hertz. The number of cycles per second.
IET	Instantaneous electronic trip. A fault condition that occurs while the drive is running insulting in a motor coast-to-stop. The drive senses a condition that could result in equipment damage and tume itself off.
Inout Power Factor	The relio of the input inverter A-C effective power to the input A-C apparent Power.
Inverter	A static power convertar used to change D-C power to A-C power.
Line Dip	A short duration power loss below the specified A C line input rating due to a power system fault.
Load torque	The motor torque required to keep the load misting at nearly constant speed.
NEMA	The abbreviation for the National Electrical Manufacture's Association
NEMA 1	The Type 1 enclosure defined in NEMA stancerds that provides protection equinat accidental or inadvertent bodity contact with live parts and a limited amount of failing cirt.
NEMA 4X	The Type 4X enclosure defined in NEMA standards that provides a degree of protection from failing rain, splashing water, hose directed water and requires a corrosion test. The Type 4X is also used indoora only.
NEMA 12	The Type 12 enclosure defined in NEMA stancards provides a degree of protection against cust, dirl, "ber, flying, dripping water and external condensation of non-corrosive liquids.
Output Frequency Hange	The frequency range allowed for continuous operation of the output wave of a drive.
Overcurrent	A current greater than a specified maximum current value.
Overload Capacily	The overload current allowed by the drive in a specified time. Overload current is indicted by the drives display as a percentage of the rated output current.
Overtemperature	A temperature greater than the specifies (rated) temperature limit.

Powar Factor	An abstract value which gives a measure of the relative proportions of resistive and reactive circuit elements. The ratio of true to apparent power.
Pulse Width Modulation	A method whereby a D-G voltage is converted to produce an A-C voltage whose magnitude and frequency can be varied.
Rafec input Voltage	Specified A-C ine vollage, at 50 or 60 Hz line irequency, connected to a drive.
Rated Output Current	Total maximum current delivered out of a drive or to a motor under full load conditions.
Rated Output Voltage	Total maximum output voltage of a drive while onlivering raced current under full load conditions.
Rated Output	Eurodamostol autor - umus feet usersor
riequency	Fundernentar output wave nequency.
Reclifier	A static power converter used to change A-C power to D-C power.
Relative Humidity	The humidity of the surrounding environment where the inverter is to be installed
Stall	A motor state where the motor remains inctionless even though the inctor is generating torque.
Surge Suppressor	Circuit protection which suppresses the peak value of any unusual input voltage to the crive, its sometimes used to lighten the leading edge of voltage.
Torque	Tendency of a motor to produce rotation (Rotational force) It is a measure of the motor to develop power.
Torque Compensation	The increase of the volts/Trequency ratio of the orive, in the low frequency area, to compensate for the reduced foreue of the motor at low spects. Heduced forque at low acceds is due to the resistance of the motor stator windings.
Voltage Feedback	A voltage signal used by the microcontroller to control the operation of the crive.
Voltage/Frequency	The ratio to output voltage (in volta) to output frequency (in hertz) in the output frequency range of the drive to achieve constant torque in the motor.
Voltage/Frequency	
(V/F) aelling	Base speed setting of the mator providing constant forque.

Voltage Regulator	A circuit which detects and uses voltage level changes to crovide a constant output voltage level.
Vallaçe to Frequency (V/F) Convertor	A circuit the, converts voltage into frequency, in the drive, this rarcuit converts the frequency reference voltage into pulses proportional to the inverter output frequency.
Werning	Alerts a person to obtantial bodily injury if the procedure is not followed.

9: Index

A

L	Page
%L()A]	4-2, 4-5, 4-7
Acceleration function (F-D1)	5-4
altitude	1-6, 2-2, 3-1
ambient temperature	2-5
analog output, metering	2-7. 3-20. 3-21, 3-23
arrow keys	4-6
atmosphere	2-5

B

base speed	5-10, 5-11
block diagram, functional	*-4
block diagram, system	7-2.7-3
branch circuit protection	3-9. 6-2

c

Carrier Frequency function (F-10)	518,61
cloarances	33, 34, 36
Coast Stop Enable function (F-16)	5-19
code, lault	7-4
cantral logic	2-6
controller running, IET. (F-09)	2-7. 5-12
Current Limit function (F-05)	5-B
18 St.	

D

Deceleration function (F 02)	5.5
default settings	5-1
diagram, Contro terminal strip	3-9
diagram, Power terminal strip	3-17
diagram, system	7-2.7-3
dimensions, physical size A	3-3
dimensions, physical size B	S-5
dimensions, physical size C	\$-7
diaconnect, input	3-8
display mode	4-1. 1-2, 4-5, 4-7,
	48.516
dynamic braking kit	2 11, 3 19, 3 22,
204 UN2222012 UN222012 - UN222012 - UN222012 UN222000000000000000000000000000000000	5-27 through 3-29

E

Electronic Thermal Overload function (F-14)	5-17
Electronic Thermal Overload Enable	
function (F-15)	5-18
elevation	2.5.3-1
emergency stop	3 24 and 3 25
error log	4-10, 5-21, 7-4

F

Page

89. W. BY	200
menory semogs	5-1
fault code	4-10.7-1
feult onde action list	7-8 7-7
formand allowed and and and and and and and and and an	2 25 4 2 4 6
IDIW2-0 GREEDD1	3-20, 4-3, 4-0
fourth deplay mode	4-2, p-1-3
fraguer cy. carrier	1-3, 5-13
frequency minimum maximum	5.8 6.6 5.7
TU10:101 K068	3-14, 3-15, 3-16, 3-20
fuses, branch circuit protection	3-11, 3-12
G	
pair: remote reference	5-14
gar remains	3.10
graunaing	2-12
n (0.0725207
IET Output Enable function (F-09)	5-12
IET relay	27.323
Indiocoan	41 69 49
Indiversity in the second seco	41, 44, 43
Input Ampa	2-1 through 2-1
input KVA	2-1 Lhrough 2-4
inputs, specifications	2-6
J	
jumper (10)	2.26
Teuber Ash	2.20
к	
keypad and display, operation	41
L	
line fractioners	4.8
0 10 000.000 10.9 · · · · · · · · · · · · · · · · · · ·	
line voltage variation	8-0
local control	5-3
location, mounting	3-1 Lhiputh 3-4, 3-8
loo arm	4.10 5.2 7.4
log. 2000	410,02,74
M	
Manual Torque Boost function (F-06)	8.9
Minimum Second function (E. 192	5.45
menuntrian estada	
maximum ioad	5-9
Maximum Speed function (F-04)	5-7
MODE/ENTER key	4-2 4-5 through 4-10.
	5.3 5.01 7.4
	0.0, 0.01, r.4
inide numbers	2-1 Cirough 2-4, 2-11,
	32
modes, display	42, 47, 4B, 49
monitor RPM	4.2
mailer	T T Have de C da
mosor	2-a phough 2-10,
	3-13
mounting, controller	2-2 through 3-7

N

Page

o

offaet, ramote reference	5-15,
output amps	2-1 .hrough 2-4
output specifications	2-7
overgurrent	5-8
overload, electronic thermal	5-17, 7-6

P

parts	7-8, 7-9
planning, Ineta lation	S-1
power lactor, displacement	2-B
powerup	47.80
PROGRAM	4-3, 4-9

R

ratings, controller	21 through 2.4
Rample Enable function (E-00)	53
Remote Reference Display Enable function	
(F-13)	5-14, 5-16
Remote Reference Gain function (F-11)	5-15, 5-16
Remote Reference Offset function (F 12)	5.15
Reverse Disable function (F-17)	5-20
R-V-RSE key	4-8
RPM at Base Speed function (F-05)	5-13
RPM monitor	4-2
RUN	43

s

scaling, olfset, gain	3-23, 5-4, 5-5
shipment	141
specifications, controller	2-1
speed, maximum	5-7
speed minimum	5-8
speed reference local remote	3-26
START	44
stop pushbatten	3-25
STOP/RESET key	4-5, 4-9, 4-10
alorage	*-6
eyelem diagram	7-2, 7-3

т

terminal strip, control	5-20
terminal strip, power	2 7
thermal overload	57
tarque baost	5-9
torque ratings	5-18, 5-21
transformer	3-11

Page

	Contraction of the second
version, software	51, 5.21
Volts/1z Base Speed function (F-07)	5-10
VOLTS	4-2, 4-7

W

٧

walls, power loss	2-12
witing, power	3-17 through 3-19
witing, control	3 20 through 3 25

Settings Table

		Factory	Bange of	Use	r Data
Function	Description	Setting	Setting	Date	Setting
F-00	Remote Enable	OFF	On, Off		
F-01	Acceleration Rate (sec)	5.0	0.5-30.0		
F-02	Deceleration Rate (sec)	5.0	1.0-30.0		
F-03	Minimum Speed (Hz)	5.0	8.0-30.0		
F-04	Maximum Speed (Hz)	60.D	30.0-240.0		
F-05	Current Limit (%)	150%	10%-150%		
F-06	Manual Torque Boost (%)	2%	0%-10%		
E-07	Volts/Hertz Base Speed	60	30-240		
F-08	RPM at Base Speed	1750	1-9999		
F 09	IET Output Enable	QFF	On, Off		
F-10	Carrier Frequency (kHz)	8	4, 6, 6		
E11	Remote Reference Gain (%)	100%	60%-100%		
F-12	Remote Reference Ofiset (%)	0%	0%-40%		
F-13	Remote Reference Display Enable	OFF	On, Off		
F-14	Electronic Thermal Overload (%)	100%	20% 100%		
F-15	Electronic Thermal Overload Enable	огг	On, Off		
F-18	Coast Stop Enable	ON	On, Off		
F-17	Reverse Disable	OFF	On. Off		
F-18	Software Version	Read Only	N/A		
ERR	Error Log follows F-18		N/A		

A-C DRIVES TRAINING AND AUDIO/VISUAL PRODUCTS

Reliance Electric affers a wide variety of inclustrial Training courses for electricians, electronic technicians and engineers who are reeponsible for the installation, repair and maintenance of production eculoment and systems.

Professional quality A/VPrograms are also available. These programs have been designed to provide years of efficient in-house training. Available for playback at the user's convenience, these videotape programs allow individual or groups to learn or review subjects at any time.

Printed reference materials come with all diagnostic and troubleshooting programs.

Training Courses

No.	Title
	A-C DRIVE COURSES
2-1	Maintenance and Troubleshooting of Wil Style Variable Speed A-C Drives
2.4	A-C Drives Hands-On Troubleshooting Lab
23	Maintenance and Troubleshooting of PW'V Style Variable Speed A C Drives
2-15	Regional Class – Maintenance and Troubleshooting of A-C Drives and Systems
4-15	Regional Class - Productive Maintenance Training

Audio/Visual Products

Order No.	Title	Format	Price
5	A-C DRIVES PROGRAMS		*
TM2240	A C Matar Theory	Videolape	\$725
TM2241	Introduction to the VVI	Videotape	725
TM2242	Troubleshooting the VVI Hegulator	Videotape	995
TM2367	Troubleshooting the 6-Transistor Power Module	Videotape	725
-	NEW VIDEO TRAINING PROGRAMS		
VMBA001	Fundamentals of A-C Motors	Videotape	\$495
VMBV201	Concepts of Digital Controls	Videolape	495
VWVSpan	GP2000 Video Training	Videotapie	495
VWVS002	LIB2000 Viceo Training	Videotape	495
VWVS005	Basics of A-C Drives	Videotape	495

For details and prices on these courses, audio/visual products and FREE Training Schedule Brochure, HD-405, contact:

> Industrial Training Department, Reliance Bectric 35000 Curtis Boulevard Castlake, Ohio 44095

Call Toll Free:

800-RELIANCE (900-735-4262)



Forward To: Reliance Flectric RGA Technical Publications 23001 Tungsten Road Cleveland, OH 44-17



Techniciae Wriding Informal Clear	
UVF #	

V+S DRIVES & INDUSTRIAL CONTROLS DOCUMENTATION IMPROVEMENT FORM

Document Number:-----

Page Number(s):-----

Comments: (Please give chapters, page numbers or specific paragraphs that the change will af tect. Include markups from the document or strach additional pages it necessary.)

What will this improvement suggestion provide?

Orginalor:	C.y:	State:	ZIP:
Company:	Phone: (
kodress:	Date:		
Tochnical Whiteg Isternal Use:	Follow-Up Action		
Witter	Date:		


Reliance Electric / 24705 Euclid Avenue / Cleveland, Ohio 44117 / 216-266-7000



Printed in U.S.A.

D2-3267-1

December, 1993